Oscillation-Bound Estimation of Perturbations Under Bandler–Kohout Subproduct

Yiming Tang^D, *Member, IEEE*, and Witold Pedrycz^D, *Life Fellow, IEEE*

Abstract—The Bandler-Kohout subproduct (BKS) method is one of the two widely acknowledged fuzzy relational inference (FRI) schemes. The previous works related to its stability and robustness mainly concentrated on how the output values were changed with perturbation parameters of input values. However, the works on estimating oscillation bounds of output values with regard to varying limits of input, are lacking. In this study, we investigate the oscillation-bound estimation of perturbations for BKS. First, the BKS output variation scopes are acquired for interval perturbation, where the R-implication, (S, N)-implication, QL-implication, and *t*-norm implication are adopted. Second, in allusion to the more sophisticated problem of the fuzzy reasoning chain with BKS, the oscillation bounds of BKS output resulting from input interval perturbation are offered. Third, we construct the upper and lower bounds of BKS output deviation originated in the simple perturbation of the input fuzzy set, in which the situations of one rule and multiple rules are both dissected. Finally, the stable properties of all these BKS strategies are confirmed. It is emphasized that interval perturbation and simple perturbation are more general ways to give expression describing the robustness issue, and the obtained oscillation bounds also deliver more detailed characterization of the output deviation along with the input perturbation. This study further validates the sound properties of the BKS method.

Index Terms—Bandler–Kohout subproduct (BKS), compositional rule of inference (CRI), fuzzy reasoning, perturbation, robustness, stability.

I. INTRODUCTION

F UZZY reasoning is embodied as an advanced computing framework on the strength of the concepts of the fuzzy set, fuzzy if-then rule, and approximate reasoning. It has been thoroughly explored and applied to the fields of fuzzy control, pattern recognition, decision making, time-series analysis

Manuscript received 27 May 2020; revised 16 August 2020; accepted 18 September 2020. Date of publication 15 January 2021; date of current version 1 July 2022. This work was supported in part by the National Natural Science Foundation of China under Grant 61673156, Grant 61877016, Grant 61672202, Grant U1613217, and Grant 61976078. This article was recommended by Associate Editor P. P. Angelov. (*Corresponding author: Yiming Tang.*)

Yiming Tang is with the Anhui Province Key Laboratory of Affective Computing and Advanced Intelligent Machine, School of Computer and Information, Hefei University of Technology, Hefei 230601, China, and also with the Department of Electrical and Computer Engineering, University of Alberta, Edmonton, AB T6R 2V4, Canada (e-mail: tym608@163.com).

Witold Pedrycz is with the Department of Electrical and Computer Engineering, University of Alberta, Edmonton, AB T6R 2V4, Canada, also with the Department of Electrical and Computer Engineering, Faculty of Engineering, King Abdulaziz University, Jeddah 21589, Saudi Arabia, and also with the Systems Research Institute, Polish Academy of Sciences, 01-447 Warsaw, Poland (e-mail: wpedrycz@ualberta.ca).

Digital Object Identifier 10.1109/TCYB.2020.3025793

and others. The general form of fuzzy reasoning comes as the following fuzzy modus ponens (FMP):

Input:
$$u$$
 is C
Rule: If u is A , then v is B
Output: D is $f(A, B, C) = ?$ (1)

in which $A, C \in F(U), B, D \in F(V)$, and F(U), F(V), respectively, represent the set of all fuzzy subsets of universes U and V. f signifies a fuzzy reasoning mechanism. The output D = f(A, B, C) is obtained from the rule and the input with the fuzzy reasoning mechanism f.

Directing at such a fuzzy reasoning problem, there are a number of reasoning mechanisms in fuzzy rule-based systems, including fuzzy relational inference (FRI) [1], similarity-based reasoning (SBR) [2], and inverse truth functional modification [3], fuzzy interpolative reasoning, and differently implicational reasoning [5], as a few representatives. Among them, FRI has gained visible attention from both theoretical and practical points of view. Among the FRI methods, the two most frequently considered strategies are the compositional rule of inference (CRI) [6] established by Zadeh, and the Bandler–Kohout subproduct (BKS) proposed by our research team [7] (realized on the basis of the previous results of Bandler and Kohout [8]). In this work, we concentrate on the FRI mechanism, especially the BKS method.

A. BKS and CRI

Here, we briefly review the BKS method. Our research team [7] came up with the BKS method of fuzzy reasoning. In allusion to a given SISO rule in (1), the BKS reasoning mode is shown as follows:

$$D = C \triangleleft R. \tag{2}$$

Thereinto, $C \in F(U)$ is the input. $R : U \times V \rightarrow [0, 1]$ means the fuzzy relation, viz., $R \in F(U \times V)$ denotes the rule base. $D \in F(V)$ represents the constructed output. In the end, the operation \triangleleft is the mapping $\triangleleft : F(U) \times F(U \times V) \rightarrow F(V)$ conveyed by

$$D(v) = \inf_{u \in U} \{ C(u) \to R(u, v) \}, \quad v \in V.$$
(3)

Here, \rightarrow represents a fuzzy implication. The mapping \triangleleft is also known as the inf -I composition, where I indicates a

2168-2267 © 2021 IEEE. Personal use is permitted, but republication/redistribution requires IEEE permission. See https://www.ieee.org/publications/rights/index.html for more information. fuzzy implication. For the rule base R, two alternatives are sought

$$R(u, v) = \bigvee_{i=1}^{n} \{A_i(u) \otimes B_i(v)\}, \quad u \in U, \quad v \in V$$
 (4)

$$\widehat{R}(u,v) = \wedge_{i=1}^{n} \{A_i(u) \to B_i(v)\}, \quad u \in U, \quad v \in V.$$
 (5)

Here, \otimes denotes a *t*-norm, and \rightarrow represents a fuzzy implication. Equation (4) corresponds to a conjunctive mode for the rule base, while (5) denotes an implicative mode. As for the difference in semantics expressed by \check{R} and \hat{R} , refer to [9]. By making use of \check{R} or \hat{R} , the corresponding BKS algorithm becomes

$$D(v) = \inf_{u \in U} \left\{ C(u) \to \bigvee_{i=1}^{n} (A_i(u) \otimes B_i(v)) \right\}, \quad v \in V$$
 (6)

$$D(v) = \inf_{u \in U} \left\{ C(u) \to \wedge_{i=1}^{n} (A_i(u) \to B_i(v)) \right\}, \quad v \in V.$$
(7)

For the case of a single rule, we can acquire two types of the BKS methods

$$D(v) = \inf_{u \in U} \{ C(u) \to (A(u) \otimes B(v)) \}, \quad v \in V$$
(8)

$$D(v) = \inf_{u \in U} \{ C(u) \to (A(u) \to B(v)) \}, \quad v \in V.$$
(9)

The BKS methods indicated by (6) or (8) are referred to as the BKS-T methods, meanwhile, the BKS methods completed with the aid of (7) or (9) are said to be the BKS-I methods.

Another classic fuzzy reasoning method is the CRI method proposed by Zadeh [6]. The CRI method can be shown as follows:

$$D(v) = \sup_{u \in U} \{ C(u) \otimes R(u, v) \}, \quad v \in V$$
(10)

where \otimes is a fuzzy conjunction, which is usually treated as a *t*-norm (see [10]).

Here, we compare the CRI method with the BKS method. Notice that \hat{R} and \hat{R} are typical and recognized fuzzy relations used to model the rule base. On the one hand, any fuzzy rule in \hat{R} is considered as a constraint, which results in a conjunctive way of integrating the individual constraints, because of more constraints, the fewer possible values satisfying these constraints [9], [11]. The employed fuzzy implication obviously expresses the logical operation IF-THEN while the minimum evidently reflects the logical conjunction, and hence \hat{R} is an appropriate structure of conjunctively merged IF-THEN fuzzy rules. On the other hand, \hat{R} is not mainly constructed as a mathematical structure of conditional IF-THEN sentences but it adopts the Cartesian product of antecedent and consequent fuzzy sets. Fuzzy rules characterized by \dot{R} are regarded as pieces of data that are evidently accumulated by the maximum in (4) for the merging of fuzzy rules [9], [11]. It is emphasized that some of the superiorities of the CRI method depend largely on the fuzzy rules modeled by \dot{R} . However, aiming at \hat{R} , many of the advantages of CRI have been lost, which incorporate the robustness of CRI in regard to the indistinguishability of input fuzzy sets in the situation of multiple rules, the equivalence of first-infer-then-aggregate (FITA), and first-aggregate-then-infer (FATI), and many strategies to improve the efficiency of the reasoning process [12]. But the BKS method has ideal performance for the fuzzy rules modeled by \hat{R} [12]. Generally speaking, in view of previous research (e.g., [12]–[14]), when a given fuzzy rule base is modeled by the fuzzy relation \check{R} , the CRI method is preferable; however, when it is modeled by the fuzzy relation \hat{R} , the BKS method is better. In addition, the fuzzy relation \hat{R} does better in handling different levels of specificity of knowledge than \check{R} , and only \hat{R} has a built-in ability for detecting inconsistencies in rule bases [15]. Therefore, for complex tasks, such as in data mining, logic programming, or complex expert systems, the combination of BKS with \hat{R} provides a better selection than the combination of CRI with \check{R} .

B. Research on Stability and Robustness for CRI and BKS

For a fuzzy reasoning mechanism, its stability and robustness are deemed as the key research topics. At present, fuzzy reasoning has become a vital strategy for designing and analyzing fuzzy controllers. The fuzzy controllers perform well when using the quantitative expression to characterize granular domain knowledge, thus the divergence of the latter from related quantitative representations enhances the importance of the stability issue for fuzzy controllers [16]. The stability of fuzzy reasoning concerns the deviation of the output implied by perturbations of the input. Current research chiefly focuses on how the output values of the reasoning strategy are changed in response to the perturbation parameters of input values.

In this regard, some studies were completed, especially for the CRI method. Pappis [17] established the proximity measure originated from approximately equal fuzzy values of system variables and relations and revealed its retention abilities in the setting of the CRI method. Then, Hong and Hwang [18] extended it to the α -similarity measure. Ying [19] presented maximum and average perturbations of fuzzy sets, determined perturbation parameters for some CRI strategies. Cai [20] made a thorough inquiry into the robustness of specific fuzzy implications and reasoning rules in CRI strategies with regard to δ -equalities of fuzzy sets. Jin *et al.* [21] offered the concept of perturbation of fuzzy sets with a logicoriented equivalence measure and obtained corresponding robustness outcomes for some fuzzy-logic connectives, fuzzy implications, reasoning rules, and fuzzy reasoning machines. Dai et al. [22] investigated the perturbation of CRI with respect to the perturbation of fuzzy sets derived from normalized Minkowski distances. Li et al. [23] adopted an idea similar to the modulus of continuity to analyze the robustness of some fuzzy connectives together with the CRI method. Li et al. [24] proposed a divergence measure on the strength of previously proposed dissimilarity function DF-metric [25] and then inquired into the robustness of fuzzy connectives and CRI. As for the BKS algorithm, there are also some works related to this aspect. Stěpnička and Javaram[12] verified the effect of the BKS method with residuated implications, which included the equivalence and reasonability for the solvability and interpolativity, and analyzed its robustness (i.e., the preservation of the indistinguishability that may be inherent in the input fuzzy sets) along analogous lines of study as [26] for CRI. Finally, it was concluded that the BKS strategy is a sound alternative as the CRI method. Henceforth, Mandal and Jayaram [13] investigated the BKS method employing Yager's classes of fuzzy implications and found that the

corresponding BKS method also has useful properties, viz., interpolativity, continuity, and robustness (whose definitions are analogous to those present in [12]). Štěpnička *et al.* [27] verified the robustness with regard to the combination of BKS and implicative model from another viewpoint, and briefly analyzed the preservation of desirable properties for FRI methods. Focusing on the satisfiability problem for the three "interpolation" axioms introduced by Moser and Navara in [28], Štěpnička and Mandal [14] further developed these axioms and discovered the corresponding conditionally firing rules as well as the related properties for the CRI and BKS algorithms with Mamdani–Assilian and implicative models.

As a matter of fact, there is another aspect of the stability of fuzzy reasoning, that is, to research how to estimate oscillation bounds of output values when variation limits of input values are provided. In [29], in allusion to the CRI method with a single rule, Cheng and Fu analyzed the upper and lower bounds of the output error induced by the simple perturbation of the input fuzzy set and determined the oscillation scope of the output result for the case of which the input ranges over an interval.

C. Motivation

It should be emphasized that the oscillation-bound estimation for BKS demands to be explored.

The previous investigation on error estimation for CRI can also be done for other FRI mechanisms. In this article, we focus on the oscillation-bound estimation for the BKS algorithm.

Some properties of the CRI mechanism rely, to a great extent, on the modeling strategy used for fuzzy rules via the fuzzy relation \check{R} , which is proper in the context in which the fuzzy rules are regarded as positive pieces of information [9], [11]. For example, the work [29] concerns the case of a single rule with \check{R} formed in CRI [viz., $R(u, v) = A(u) \rightarrow B(v)$]. However, there are also situations in which the conditional aspect of fuzzy rules is considered, and the fuzzy relation \hat{R} has to be employed to characterize them [15]. Hence, some properties of CRI cannot hold, including the oscillation-bound estimation. As a consequence, the oscillation-bound estimation of BKS has to be formed for both \check{R} and \hat{R} .

D. Outline

Section II covers some preliminaries. Section III discusses the estimation of the limits of the output with regard to the interval perturbation of the input. Section IV analyzes the estimates of the output scope related to the interval perturbation of the input in the problem of a chain of fuzzy reasoning via BKS. Section V investigates the error estimation of perturbations for BKS with a single rule and with multiple rules. The stability of the BKS fuzzy reasoning method is validated. Section VI offers some discussion. Section VII summarizes this article.

II. PRELIMINARIES

A. Some Notions

There are several definitions of fuzzy implications. The fundamental one is as follows.

Definition 1 [8], [30]: A fuzzy implication on [0, 1] is a function $I : [0, 1]^2 \rightarrow [0, 1]$ satisfying the condition

(P1)
$$I(0,0) = I(0,1) = I(1,1) = 1$$
, $I(1,0) = 0$.

I(a, b) can also be be represented $a \rightarrow b$ $(a, b \in [0, 1])$.

Definition 2 [10]: A function \otimes : $[0, 1]^2 \longrightarrow [0, 1]$ is referred to as a *t*-norm if \otimes is associative, commutative, increasing, and lets $1 \otimes p = p$ be valid $(p \in [0, 1])$.

Here, three commonly used *t*-norms are adopted, which incorporate min \otimes_M , product \otimes_P , and the Lukasiewicz conjunction \otimes_L . That is, $\otimes_M(p,q) = p \land q$, $\otimes_P(p,q) = p \times q$, and $\otimes_L(p,q) = 0 \lor (p+q-1)$ where $p, q \in [0, 1]$ and \lor and \land signify the supremum and infimum in turn.

Definition 3 [10]: A function \oplus : $[0, 1]^2 \longrightarrow [0, 1]$ goes by the name of a *t*-conorm if \oplus is associative, commutative, increasing, and makes $0 \oplus p = p$ be effective ($p \in [0, 1]$).

Definition 4 [10]: A fuzzy negation is a decreasing function $N : [0, 1] \rightarrow [0, 1]$ which meets N(0) = 1 and N(1) = 0.

Definition 5 [30]: A fuzzy implication \rightarrow is referred to as an *R*-implication whenever a left-continuous *t*-norm \otimes exists and lets $p \rightarrow q = \lor \{x \in [0, 1] | p \otimes x \le q\}$ be right $(p, q \in [0, 1])$.

Definition 6 [31]: A function $I : [0, 1]^2 \rightarrow [0, 1]$ is known as an (S, N)-implication whenever there exist a *t*-conorm \oplus and a fuzzy negation N making $I(p, q) = N(p) \oplus q$ hold $(p, q \in [0, 1])$. Such (S, N)-implication is recorded as $I_{\oplus,N}$.

Definition 7 [30], [32]: A function $I : [0, 1]^2 \rightarrow [0, 1]$ is referred to as a QL-implication whenever there are a *t*-norm \otimes , a *t*-conorm \oplus , and a fuzzy negation N making I(p, q) = $N(p) \oplus (p \otimes q) \ (p, q \in [0, 1])$ be effective. Such QL-implication is represented by $I_{\otimes, \oplus, N}$.

Definition 8 [29], [33]: A function $I : [0, 1]^2 \rightarrow [0, 1]$ is referred to as a *t*-norm implication whenever there exists a *t*-norm \otimes making $I(p, q) = p \otimes q$ be right $(p, q \in [0, 1])$.

Although the *t*-norm implications do not meet (P1), they are adopted as a model of fuzzy implication in many applications of fuzzy logic [33].

Definition 9 [17]: Suppose that $C, D \in F(Z)$ and that $\lambda \in [0, 1]$. If $\sup_{z \in Z} |C(z) - D(z)| \le \lambda$ holds, then C and D are said to be approximately equal. λ is known as a proximity measure of C and D.

Definition 10 [18]: Suppose that $C, D \in F(Z)$ and that $\lambda \in [0, 1]$. If $1 - \sup_{z \in Z} |C(z) - D(z)| \ge \lambda$ holds, then C and D are said to be λ -similar.

Definition 11 [19]: Suppose that $C, D \in F(Z)$ and that $\lambda \in [0, 1]$. If one has $|C(z) - D(z)| \leq \lambda$ for any $z \in Z$, viz., $D(z) = C(z) + \lambda \psi(z)$ for some function $\psi : Z \rightarrow [-1, 1]$ (where $0 \leq C(z) + \lambda \psi(z) \leq 1$ for any $z \in Z$), then *D* is referred to as a maximum λ perturbation of *C*.

Definition 12 [34]: Suppose that $C, D \in F(Z)$ and that $\lambda \in [0, 1]$. If $\sup_{z \in Z} |C(z) - D(z)| \le 1 - \lambda$ holds, then C and D are called λ -equal.

Definition 13 [29]:

- 1) Suppose that $\beta^-, \beta^+ \in F(Z)$ and that $\beta^-(z) \leq \beta^+(z)$ $(z \in Z)$, then $[\beta^-, \beta^+]$ is known as a fuzzy interval on Z.
- 2) Suppose that $C \in F(Z)$ and that $[\beta^-, \beta^+]$ be a fuzzy interval on Z. If $\beta^-(z) \le C(z) \le \beta^+(z)$ ($z \in Z$), then

 $[\beta^-, \beta^+]$ is referred to as an interval perturbation of *C*, which is signified by $C \in [\beta^-, \beta^+]$.

Definition 14 [29]: Suppose that $A, B \in F(Z)$. If there is a function $\beta : Z \rightarrow [-1, 1]$ making $B(z) = A(z) + \beta(z)$ ($z \in Z$), then *B* is known as a simple perturbation of *A*, and β is called a perturbation factor of *A*.

Definition 15 [29]: Assume that $C, E \in F(Z), D \in F(W)$, and that $C \in [\alpha^-, \alpha^+], D \in [\eta^-, \eta^+], E \in [\gamma^-, \gamma^+]$. A fuzzy reasoning strategy f is referred to as a stable function if, for any $\varepsilon > 0$, there is a fuzzy interval $[\lambda^-, \lambda^+]$ on W and $\delta > 0$ making $\lambda^+(v) - \lambda^-(v) < \varepsilon$ be effective for any $w \in W$ and $f(C, D, E) \in [\lambda^-, \lambda^+]$ if $\alpha^+(z) - \alpha^-(z) < \delta, \eta^+(w) - \eta^-(w) < \delta, \gamma^+(z) - \gamma^-(z) < \delta$ ($z \in Z, w \in W$).

Definition 16 [29]:

- 1) Assume that $C, E \in F(Z)$, $D \in F(W)$, and that C^* , D^* , E^* are perturbations of C, D, E with factors $\beta_1, \beta_2, \beta_3$ in turn. A fuzzy reasoning strategy f is known as a stable function if, for any $\varepsilon > 0$, there is $\delta > 0$ making $|f(C^*, D^*, E^*)(w) - f(C, D, E)(w)| < \varepsilon$ hold if $|\beta_1(z)| < \delta, |\beta_2(w)| < \delta, |\beta_3(z)| < \delta (z \in Z, w \in W).$
- 2) Assume that $\{C_m^*, D_m^*, E_m^*\}$ is a perturbation sequence of (C, D, E) with regard to $\{\beta_{1m}, \beta_{2m}, \beta_{3m}\}$ of factors (m = 1, 2, ...). Then, a fuzzy reasoning strategy f is referred to as an asymptotic stable function if, for any $\varepsilon > 0$, there are m_0 and $\delta > 0$ making $|f(C_m^*, D_m^*, E_m^*)(w) f(C, D, E)(w)| < \varepsilon$ hold for any $m > m_0$ if $|\beta_{1m}(z)| < \delta$, $|\beta_{2m}(w)| < \delta$, and $|\beta_{3m}(z)| < \delta$ $(z \in Z, w \in W)$.

Lemma 1 [29]: Let h_1 and h_2 be real-valued, bounded mappings on Z, and $C, D \in F(Z)$. Thus, one has:

- 1) $h_1 \vee h_2 = \max\{h_1, h_2\} = (h_1 + h_2)/2 + |h_1 h_2|/2;$
- 2) $h_1 \wedge h_2 = \min\{h_1, h_2\} = (h_1 + h_2)/2 |h_1 h_2|/2;$
- 3) $-|h_1 h_2| \le |h_1| |h_2| \le |h_1 h_2|;$
- 4) $\inf_{z \in Z} C(z) + \inf_{z \in Z} D(z) \le \inf_{z \in Z} (C(z) + D(z));$
- 5) $\sup_{z \in \mathbb{Z}} (C(z) + D(z)) \leq \sup_{z \in \mathbb{Z}} C(z) + \sup_{z \in \mathbb{Z}} D(z);$
- 6) $\sup_{z \in Z} C(z) = -\inf_{z \in Z} (-C(z));$
- 7) $\inf_{z \in Z} (C(z) \wedge D(z)) = \inf_{z \in Z} C(z) \wedge \inf_{z \in Z} D(z);$
- 8) $\inf_{z \in Z} (C(z) \lor D(z)) \ge \inf_{z \in Z} C(z) \lor \inf_{z \in Z} D(z);$
- 9) $\sup_{z \in Z} (C(z) \wedge D(z)) \leq \sup_{z \in Z} C(z) \wedge \sup_{z \in Z} D(z);$
- 10) $\sup_{z \in Z} (C(z) \lor D(z)) = \sup_{z \in Z} C(z) \lor \sup_{z \in Z} D(z).$
- It is effortless to verify Lemma 2.

Lemma 2: If h_1 and h_2 be real-valued, bounded mappings on Z, then:

- 1) $\sup_{z \in Z} h_1(z) \sup_{z \in Z} h_2(z) \le \sup_{z \in Z} (h_1(z) h_2(z));$
- 2) $\inf_{z \in Z} h_1(z) \inf_{z \in Z} h_2(z) \le \sup_{z \in Z} (h_1(z) h_2(z));$
- 3) $\inf_{z \in Z} h_1(z) \inf_{z \in Z} h_2(z) \ge \inf_{z \in Z} (h_1(z) h_2(z));$
- 4) $\sup_{z \in Z} h_1(z) \sup_{z \in Z} h_2(z) \ge \inf_{z \in Z} (h_1(z) h_2(z)).$

III. INTERVAL PERTURBATION FOR BKS

A. Interval Perturbation for BKS With Single Rule

Assume that $[\alpha^-, \alpha^+]$ and $[\gamma^-, \gamma^+]$ are fuzzy intervals on U, and that $[\eta^-, \eta^+]$ is a fuzzy interval on V. We denote $\tilde{i}_u(f) = \inf_{u \in U} \{f(u)\}, \tilde{s}_u(f) = \sup_{u \in U} \{f(u)\}, \text{ and } BKS_1(v) = BKS(A, B, C)(v) \ (v \in V).$

Theorem 1: Let $A, C \in F(U)$ and $B \in F(V)$. Assume that $A \in [\alpha^{-}, \alpha^{+}], C \in [\gamma^{-}, \gamma^{+}]$, and $B \in [\eta^{-}, \eta^{+}]$, and that \otimes is a *t*-norm. The BKS-T method is adopted.

- 1) If \rightarrow is an R-implication, then $\tilde{s}_u(\gamma^+) \triangleleft (\tilde{i}_u(\alpha^-) \otimes \eta^-(v)) \leq BKS_1(v) \leq \tilde{i}_u(\gamma^-) \triangleleft (\tilde{s}_u(\alpha^+) \otimes \eta^+(v)).$
- 2) If \rightarrow is a *t*-norm implication, then $\tilde{i}_u(\gamma^-) \triangleleft (\tilde{i}_u(\alpha^-) \otimes \eta^-(v)) \leq BKS_1(v) \leq \tilde{s}_u(\gamma^+) \triangleleft (\tilde{s}_u(\alpha^+) \otimes \eta^+(v)).$
- 3) If \rightarrow is an (S, N)-implication $I_{\oplus_1,N}$, then $\tilde{i}_u\{N(\tilde{s}_u(\gamma^+)) \oplus_1 (\tilde{i}_u(\alpha^-) \otimes \eta^-(v))\} \leq BKS_1(v) \leq \tilde{i}_u\{N(\tilde{i}_u(\gamma^-)) \oplus_1 (\tilde{s}_u(\alpha^+) \otimes \eta^+(v))\}.$
- 4) If \rightarrow is an QL-implication $I_{\otimes_1,\oplus_1,N}$, then $\tilde{i}_u\{N(\tilde{s}_u(\gamma^+)) \oplus_1 [\tilde{i}_u(\gamma^-) \otimes_1 (\tilde{i}_u(\alpha^-) \otimes \eta^-(v))]\} \leq BKS_1(v) \leq \tilde{i}_u\{N(\tilde{i}_u(\gamma^-)) \oplus_1 [\tilde{s}_u(\gamma^+) \otimes_1 (\tilde{s}_u(\alpha^+) \otimes \eta^+(v))]\}.$

Proof: Here, we only prove 1). The other cases can be gained in an analogous manner. Note that an *R*-implication is decreasing in the first variable and increasing in the second variable. From the intrinsic characteristics of *R*-implications and *t*-norms, one has $\tilde{s}_u(\gamma^+) \triangleleft (\tilde{i}_u(\alpha^-) \otimes \eta^-(v)) = \inf_{u \in U} \{\sup_{u \in U}(\gamma^+(u)) \rightarrow (\inf_{u \in U}(\alpha^-(u)) \otimes \eta^-(v))\} \leq BKS_1(v) = \inf_{u \in U} \{C(u) \rightarrow (A(u) \otimes B(v))\} \leq \inf_{u \in U} \{\inf_{u \in U}(\gamma^-(u)) \rightarrow (\sup_{u \in U}(\alpha^+(u)) \otimes \eta^+(v))\} = \tilde{i}_u(\gamma^-) \triangleleft (\tilde{s}_u(\alpha^+) \otimes \eta^+(v)).$

Theorem 2 can be derived in a similar manner.

Theorem 2: Let $A, C \in F(U)$ and $B \in F(V)$. Assume that $A \in [\alpha^{-}, \alpha^{+}]$, and $C \in [\gamma^{-}, \gamma^{+}]$, $B \in [\eta^{-}, \eta^{+}]$. The BKS-I method is used.

- 1) If \rightarrow is an *R*-implication, then $\tilde{s}_u(\gamma^+) \triangleleft (\tilde{s}_u(\alpha^+) \rightarrow \eta^-(v)) \leq BKS_1(v) \leq \tilde{i}_u(\gamma^-) \triangleleft (\tilde{i}_u(\alpha^-) \rightarrow \eta^+(v)).$
- 2) If \rightarrow is a *t*-norm implication, then $\tilde{i}_u(\gamma^-) \triangleleft (\tilde{i}_u(\alpha^-) \rightarrow \eta^-(v)) \leq BKS_1(v) \leq \tilde{s}_u(\gamma^+) \triangleleft (\tilde{s}_u(\alpha^+) \rightarrow \eta^+(v)).$
- 3) If \rightarrow is an (S, N)-implication $I_{\oplus_1,N}$, then $\tilde{i}_u\{N(\tilde{s}_u(\gamma^+)) \oplus_1 (N(\tilde{s}_u(\alpha^+)) \oplus_1 \eta^-(v))\} \leq BKS_1(v) \leq \tilde{i}_u\{N(\tilde{i}_u(\gamma^-)) \oplus_1 (N(\tilde{i}_u(\alpha^-)) \oplus_1 \eta^+(v))\}.$
- 4) If \rightarrow is an QL-implication $I_{\otimes_1,\oplus_1,N}$, then $\tilde{i}_u\{N(\tilde{s}_u(\gamma^+))\oplus_1 [\tilde{i}_u(\gamma^-)\otimes_1 (N(\tilde{s}_u(\alpha^+))\oplus_1 (\tilde{i}_u(\alpha^-)\otimes_1 \eta^-(v)))]\} \leq BKS_1(v) \leq \tilde{i}_u\{N(\tilde{i}_u(\gamma^-))\oplus_1 [\tilde{s}_u(\gamma^+)\otimes_1 (N(\tilde{i}_u(\alpha^-))\oplus_1 (\tilde{s}_u(\alpha^+)\otimes_1 \eta^+(v)))]\}.$

If the generic operations (including *t*-norm, *t*-conorm, and fuzzy negation) are continuous, the BKS method with \otimes (some *t*-norm) and an (*S*, *N*)-implication, or a QL-implication, or a *t*-norm implication is stable. As for the situation of *R*-implication, if the *R*-implication and *t*-norm are continuous, then the corresponding BKS fuzzy reasoning method is stable.

Suppose that there is a perturbation sequence $([\alpha_m^-, \alpha_m^+], [\eta_m^-, \eta_m^+], [\gamma_m^-, \gamma_m^+])$ of the input (A, B, C) for the BKS method implying that $\lim_{m\to\infty} \sup_{u\in U}(\alpha_m^+(u) - \alpha_m^-(u)) = \lim_{m\to\infty} \sup_{v\in V}(\eta_m^+(v) - \eta_m^-(v)) = \lim_{m\to\infty} \sup_{u\in U}(\gamma_m^+(u) - \gamma_m^-(u)) = 0$ hold (m = 1, 2, ...). In Theorems 1 and 2, we adopt λ_m^-, λ_m^+ to indicate the corresponding lower and upper bounds of the BKS output, viz., $BKS(A, B, C) \in [\lambda_m^-, \lambda_m^+]$ for $([\alpha_m^-, \alpha_m^+], [\eta_m^-, \eta_m^+], [\gamma_m^-, \gamma_m^+])$ (m = 1, 2, ...). When the continuous condition are met, one has from Theorems 1 and 2 that $\lim_{m\to\infty} \sup_{v\in V}(\lambda_m^+(v) - \lambda_m^-(v)) = 0$. That is, the output of BKS algorithm converges steadily to a value when the continuous conditions mentioned above is effective. From Definition 15, the BKS algorithm for the situation of one rule is stable from the perspective of interval perturbation.

B. Interval Perturbation for BKS With Multiple Rules

Denote $BKS_n(v) = BKS(A_1, ..., A_n, B_1, ..., B_n, C)(v)$ ($v \in V$). Theorems 3 and 4 can be proved analogously.

Theorem 3: Suppose that $A_i, C \in F(U)$, $B_i \in F(V)$, and that $A_i \in [\alpha_i^-, \alpha_i^+]$, and $C \in [\gamma^-, \gamma^+]$, $B_i \in [\eta_i^-, \eta_i^+]$, and that \otimes is a *t*-norm (i = 1, 2, ..., n). The BKS-T method is adopted.

- 1) If \rightarrow is an *R*-implication, then $\tilde{s}_u(\gamma^+) \triangleleft [\lor_{i=1}^n (\tilde{i}_u(\alpha_i^-) \otimes \eta_i^-(v))] \leq BKS_n(v) \leq \tilde{i}_u(\gamma^-) \triangleleft [\lor_{i=1}^n (\tilde{s}_u(\alpha_i^+) \otimes \eta_i^+(v))].$
- 2) If \rightarrow is a *t*-norm implication, then $\tilde{i}_u(\gamma^-) \triangleleft [\lor_{i=1}^n (\tilde{i}_u(\alpha_i^-) \otimes \eta_i^-(v))] \leq BKS_n(v) \leq \tilde{s}_u(\gamma^+) \triangleleft [\lor_{i=1}^n (\tilde{s}_u(\alpha_i^+) \otimes \eta_i^+(v))].$
- 3) If \rightarrow is an (S, N)-implication $I_{\oplus_1,N}$, then $\tilde{i}_u\{N(\tilde{s}_u(\gamma^+)) \oplus_1 \lor_{i=1}^n (\tilde{i}_u(\alpha_i^-) \otimes \eta_i^-(v))\} \leq BKS_n(v) \leq \tilde{i}_u\{N(\tilde{i}_u(\gamma^-)) \oplus_1 \lor_{i=1}^n (\tilde{s}_u(\alpha_i^+) \otimes \eta_i^+(v))\}.$ 4) If \rightarrow is a QL-implication $I_{\otimes_1,\oplus_1,N}$, then
- 4) If \rightarrow is a QL-implication $I_{\otimes_1,\oplus_1,N}$, then $\tilde{i}_u\{N(\tilde{s}_u(\gamma^+)) \oplus_1 [\tilde{i}_u(\gamma^-) \otimes_1 \lor_{i=1}^n (\tilde{i}_u(\alpha_i^-) \otimes \eta_i^-(v))]\} \leq BKS_n(v) \leq \tilde{i}_u\{N(\tilde{i}_u(\gamma^-)) \oplus_1 [\tilde{s}_u(\gamma^+) \otimes_1 \lor_{i=1}^n (\tilde{s}_u(\alpha_i^+) \otimes \eta_i^+(v))]\}.$

Theorem 4: Suppose that $A_i, C \in F(U)$, $B_i \in F(V)$, and that $A_i \in [\alpha_i^-, \alpha_i^+]$, $C \in [\gamma^-, \gamma^+]$, and $B_i \in [\eta_i^-, \eta_i^+]$ (i = 1, 2, ..., n). The BKS-I method is utilized.

- 1) If \rightarrow is an *R*-implication, then $\tilde{s}_u(\gamma^+) \triangleleft [\wedge_{i=1}^n (\tilde{s}_u(\alpha_i^+) \rightarrow \eta_i^-(v))] \leq BKS_n(v) \leq \tilde{i}_u(\gamma^-) \triangleleft [\wedge_{i=1}^n (\tilde{i}_u(\alpha_i^-) \rightarrow \eta_i^+(v))].$
- 2) If \rightarrow is a *t*-norm implication, then $\tilde{i}_u(\gamma^-) \triangleleft [\wedge_{i=1}^n (\tilde{i}_u(\alpha_i^-) \rightarrow \eta_i^-(v))] \leq BKS_n(v) \leq \tilde{s}_u(\gamma^+) \triangleleft [\wedge_{i=1}^n (\tilde{s}_u(\alpha_i^+) \rightarrow \eta_i^+(v))].$
- 3) If \rightarrow is an (S, N)-implication $I_{\oplus_1,N}$, then $\tilde{i}_u\{N(\tilde{s}_u(\gamma^+)) \oplus_1 \land_{i=1}^n (N(\tilde{s}_u(\alpha_i^+)) \oplus_1 \eta_i^-(v))\} \leq BKS_n(v) \leq \tilde{i}_u\{N(\tilde{i}_u(\gamma^-)) \oplus_1 \land_{i=1}^n (N(\tilde{i}_u(\alpha_i^-)) \oplus_1 \eta_i^+(v))\}.$
- 4) If \rightarrow is an QL-implication $I_{\otimes_1,\oplus_1,N}$, then $\tilde{i}_u\{N(\tilde{s}_u(\gamma^+)) \oplus_1 [\tilde{i}_u(\gamma^-) \otimes_1 \wedge_{i=1}^n (N(\tilde{s}_u(\alpha_i^+)) \oplus_1 (\tilde{i}_u(\alpha_i^-) \otimes_1 \eta_i^-(v)))]\} \leq BKS_n(v) \leq \tilde{i}_u\{N(\tilde{i}_u(\gamma^-)) \oplus_1 [\tilde{s}_u(\gamma^+) \otimes_1 \wedge_{i=1}^n (N(\tilde{i}_u(\alpha_i^-)) \oplus_1 (\tilde{s}_u(\alpha_i^+) \otimes_1 \eta_i^+(v)))]\}.$

On the strength of Theorems 3 and 4, if the operations (i.e., *t*-norm, *t*-conorm, and fuzzy negation) are continuous, then the BKS method for multiple rules is stable for the case of (S, N)-implication, or QL-implication, or *t*-norm implication with \otimes . Moreover, if the *R*-implication and *t*-norm are continuous, then the BKS method for multiple rules is stable.

Assume that there is a perturbation sequence $([\alpha_{im}^{-}, \alpha_{im}^{+}], [\eta_{im}^{-}, \eta_{im}^{+}], [\gamma_{im}^{-}, \gamma_{im}^{+}])$ of the input $(A_1, \ldots, A_n, B_1, \ldots, B_n, C)$ for the BKS method making $\lim_{m \to \infty} \sup_{u \in U} (\alpha_{im}^+(u) - \alpha_{im}^-(u)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} (\eta_{im}^-(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} (\eta_{im}^-(v)) = \lim_{m \to \infty$ $\eta_{im}^{-}(v)$ = $\lim_{m\to\infty} \sup_{u\in U}(\gamma_{im}^{+}(u) - \gamma_{im}^{-}(u)) = 0$ hold (i = 1, 2, ..., n). In Theorems 3 and 4, we analogously make use of λ_m^- and λ_m^+ to signify the homologous lower and upper bounds of the BKS output. When the continuous condition is effective, we gain from Theorems 3 and 4 that $\lim_{m\to\infty} \sup_{v\in V} (\lambda_m^+(v) - \lambda_m^-(v)) = 0$. That is, the output of BKS method for multiple rules converges to a value with meeting the continuity. On the grounds of Definition 15, the BKS algorithm for the situation of multiple rules is stable from the viewpoint of interval perturbation.

IV. INTERVAL PERTURBATION FOR BKS CHAIN INFERENCE

A meaningful generalization of BKS is a chain of fuzzy reasoning. Let $U_1, U_2, \ldots, U_{n+1}$ be n + 1 universes. $C, A_1 \in$ $F(U_1), B_1, D_1, A_2 \in F(U_2), \ldots$, and finally $B_n, D_n \in$ $F(U_{n+1})$. We probe into the following reasoning chain:

Input: C						
Rule 1: A_1	implies B_1					
Output 1:	D_1					
Rule 2:	A_2	implies	B_2			
Output 2:			D_2			
Output n-1:				D_{n-}	-1	
Rule n:				A_n	implies	B_n
Final output	:					D_n .

Theorem 5: Assume that $A_i \in [\alpha_i^-, \alpha_i^+]$, $B_i \in [\eta_i^-, \eta_i^+]$, and $C \in [\gamma_1^-, \gamma_1^+]$ (i = 1, 2, ..., n), and that \otimes is a *t*-norm. The BKS-T method is adopted.

- 1) If \rightarrow is an *R*-implication, then $\tilde{s}_{u_n}(\gamma_n^+) \triangleleft_n (\tilde{i}_{u_n}(\alpha_n^-) \otimes \eta_n^-(u_{n+1})) \leq D_n(u_{n+1}) \leq \tilde{i}_{u_n}(\gamma_n^-) \triangleleft_n (\tilde{s}_{u_n}(\alpha_n^+) \otimes \eta_n^+(u_{n+1}))$, in which $\gamma_{i+1}^+(u_{i+1}) \triangleq \tilde{i}_{u_i}(\gamma_i^-) \triangleleft_i (\tilde{s}_{u_i}(\alpha_i^+) \otimes \eta_i^+(u_{i+1})), \gamma_{i+1}^-(u_{i+1}) \triangleq \tilde{s}_{u_i}(\gamma_i^+) \triangleleft_i (\tilde{i}_{u_i}(\alpha_i^-) \otimes \eta_i^-(u_{i+1}))$ $(i = 1, 2, \dots, n-1).$
- 2) If \rightarrow is a *t*-norm implication, then $\tilde{i}_{u_n}(\gamma_n^-) \triangleleft_n(\tilde{i}_{u_n}(\alpha_n^-) \otimes \eta_n^-(u_{n+1})) \leq D_n(u_{n+1}) \leq \tilde{s}_{u_n}(\gamma_n^+) \triangleleft_n(\tilde{s}_{u_n}(\alpha_n^+) \otimes \eta_n^+(u_{n+1}))$, in which $\gamma_{i+1}^+(u_{i+1}) \triangleq \tilde{s}_{u_i}(\gamma_i^+) \triangleleft_i(\tilde{s}_{u_i}(\alpha_i^+) \otimes \eta_i^+(u_{i+1}))$, $\gamma_{i+1}^-(u_{i+1}) \triangleq \tilde{i}_{u_i}(\gamma_i^-) \triangleleft_i(\tilde{i}_{u_i}(\alpha_i^-) \otimes \eta_i^-(u_{i+1}))$ $(i = 1, 2, \dots, n-1).$
- 3) If \rightarrow is an (S, N)-implication $I_{\oplus_1,N}$, then $\tilde{i}_{u_n}\{N(\tilde{s}_{u_n}(\gamma_n^+))\oplus_1(\tilde{i}_{u_n}(\alpha_n^-)\otimes\eta_n^-(u_{n+1}))\} \leq D_n(u_{n+1}) \leq \tilde{i}_{u_n}\{N(\tilde{i}_{u_n}(\gamma_n^-))\oplus_1(\tilde{s}_{u_n}(\alpha_n^+)\otimes\eta_n^+(u_{n+1}))\}, \text{ in which}$ $\gamma_{i+1}^+(u_{i+1}) \triangleq \tilde{i}_{u_i}\{N(\tilde{i}_{u_i}(\gamma_i^-))\oplus_1(\tilde{s}_{u_i}(\alpha_i^+)\otimes\eta_i^+(u_{i+1}))\},$ $\gamma_{i-1}^{-1}(u_{i+1}) \triangleq \tilde{i}_{u_i}\{N(\tilde{s}_{u_i}(\gamma_i^+))\oplus_1(\tilde{i}_{u_i}(\alpha_i^-)\otimes\eta_i^-(u_{i+1}))\}$ $(i = 1, 2, \dots, n-1).$
- 4) If \rightarrow is a QL-implication $I_{\otimes_1,\oplus_1,N}$, then $\tilde{i}_{u_n}\{N(\tilde{s}_{u_n}(\gamma_n^+))\oplus_1[\tilde{i}_{u_n}(\gamma_n^-)\otimes_1(\tilde{i}_{u_n}(\alpha_n^-)\otimes\eta_n^-(u_{n+1}))]\} \leq D_n(u_{n+1}) \leq \tilde{i}_{u_n}\{N(\tilde{i}_{u_n}(\gamma_n^-))\oplus_1[\tilde{s}_{u_n}(\gamma_n^+)\otimes_1(\tilde{s}_{u_n}(\alpha_n^+)\otimes\eta_n^+(u_{n+1}))]\}$, in which $\gamma_{i+1}^+(u_{i+1}) \triangleq \tilde{i}_{u_i}\{N(\tilde{i}_{u_i}(\gamma_i^-))\oplus_1[\tilde{s}_{u_i}(\gamma_i^+)\otimes_1(\tilde{s}_{u_i}(\alpha_i^+)\otimes\eta_i^+(u_{i+1}))]\}$, $\gamma_{i+1}^-(u_{i+1}) \triangleq \tilde{i}_{u_i}\{N(\tilde{s}_{u_i}(\gamma_i^+))\oplus_1[\tilde{i}_{u_i}(\gamma_i^-)\otimes_1(\tilde{i}_{u_i}(\alpha_i^-)\otimes\eta_i^-(u_{i+1}))]\}$ $(i = 1, 2, \dots, n-1).$

Proof: Here, we merely validate 1). It is analogous to 1) of Theorem 1 that $\tilde{s}_{u_1}(\gamma_1^+) \triangleleft_1 (\tilde{i}_{u_1}(\alpha_1^-) \otimes \eta_1^-(u_2)) \leq D_1(u_2) = BKS(A_1, B_1, C)(u_2) \leq \tilde{i}_{u_1}(\gamma_1^-) \triangleleft_1 (\tilde{s}_{u_1}(\alpha_1^+) \otimes \eta_1^+(u_2)).$

We denote $\gamma_2^+(u_2) \triangleq \tilde{i}_{u_1}(\gamma_1^-) \triangleleft_1 (\tilde{s}_{u_1}(\alpha_1^+) \otimes \eta_1^+(u_2)) =$ $\inf_{u_1 \in U_1} \{\inf_{u_1 \in U_1}(\gamma_1^-(u_1)) \to (\sup_{u_1 \in U_1}(\alpha_1^+(u_1)) \otimes \eta_1^+(u_2))\},$ and $\gamma_2^-(u_2) \triangleq \tilde{s}_{u_1}(\gamma_1^+) \triangleleft_1 (\tilde{i}_{u_1}(\alpha_1^-) \otimes \eta_1^-(u_2)) =$ $\inf_{u_1 \in U_1} \{\sup_{u_1 \in U_1}(\gamma_1^+(u_1)) \to (\inf_{u_1 \in U_1}(\alpha_1^-(u_1)) \otimes \eta_1^-(u_2))\}.$ Then, we obtain $\tilde{s}_{u_2}(\gamma_2^+) \triangleleft_2 (\tilde{i}_{u_2}(\alpha_2^-) \otimes \eta_2^-(u_3)) \leq D_2(u_3) =$ $BKS(A_2, B_2, D_1)(u_3) \leq \tilde{i}_{u_2}(\gamma_2^-) \triangleleft_2 (\tilde{s}_{u_2}(\alpha_2^+) \otimes \eta_2^+(u_3)).$

More generally, we can acquire $\tilde{s}_{u_{i+1}}$ $(\gamma_{i+1}^+) \triangleleft_{i+1}$ $(\tilde{i}_{u_{i+1}})$ $(\alpha_{i+1}^-) \otimes \eta_{i+1}^-(u_{i+2})) \leq D_{i+1}(u_{i+2}) = BKS \ (A_{i+1}, B_{i+1}, D_i)$ $(u_{i+2}) \leq \tilde{i}_{u_{i+1}} \ (\gamma_{i+1}^-) \triangleleft_{i+1} \ (\tilde{s}_{u_{i+1}}(\alpha_{i+1}^+) \otimes \eta_{i+1}^+(u_{i+2})) \ (i = 0)$ 1, 2, ..., *n*-1), in which $\gamma_{i+1}^+(u_{i+1}) \triangleq \tilde{i}_{u_i}(\gamma_i^-) \triangleleft_i(\tilde{s}_{u_i}(\alpha_i^+) \otimes \eta_i^+(u_{i+1})) = \inf_{u_i \in U_i} \{\inf_{u_i \in U_i}(\gamma_i^-(u_i)) \to (\sup_{u_i \in U_i}(\alpha_i^+(u_i)) \otimes \eta_i^+(u_{i+1}))\}, \text{ and } \gamma_{i+1}^-(u_{i+1}) \triangleq \tilde{s}_{u_i}(\gamma_i^+) \triangleleft_i(\tilde{i}_{u_i}(\alpha_i^-) \otimes \eta_i^-(u_{i+1})) = \inf_{u_i \in U_i} \{\sup_{u_i \in U_i}(\gamma_i^+(u_i)) \to (\inf_{u_i \in U_i}(\alpha_i^-(u_i)) \otimes \eta_i^-(u_{i+1}))\}.$

Finally, we gain $\tilde{s}_{u_n}(\gamma_n^+) \triangleleft_n (\tilde{i}_{u_n}(\alpha_n^-) \otimes \eta_n^-(u_{n+1})) \leq D_n(u_{n+1}) \leq \tilde{i}_{u_n}(\gamma_n^-) \triangleleft_n (\tilde{s}_{u_n}(\alpha_n^+) \otimes \eta_n^+(u_{n+1})).$

Theorem 6 is proved in a similar way.

Theorem 6: Assume that $A_i \in [\alpha_i^-, \alpha_i^+], B_i \in [\eta_i^-, \eta_i^+]$, and $C \in [\gamma_1^-, \gamma_1^+]$ (i = 1, 2, ..., n). The BKS-I method is utilized. 1) If \rightarrow is an *R*-implication, then $\tilde{s}_{u_n}(\gamma_n^+) \triangleleft_n (\tilde{s}_{u_n}(\alpha_n^+) \rightarrow \eta_n^-(u_{n+1})) \leq D_n(u_{n+1}) \leq \tilde{i}_{u_n}(\gamma_n^-) \triangleleft_n (\tilde{i}_{u_n}(\alpha_n^-) \rightarrow \eta_n^-(u_{n+1}))$

- $\eta_n^+(u_{n+1})), \text{ in which } \gamma_{i+1}^+(u_{i+1}) \triangleq \tilde{i}_{u_i}(\gamma_i^-) \triangleleft_i(\tilde{i}_{u_i}(\alpha_i^-) \rightarrow \eta_i^+(u_{i+1})) \text{ and } \gamma_{i+1}^-(u_{i+1}) \triangleq \tilde{s}_{u_i}(\gamma_i^+) \triangleleft_i(\tilde{s}_{u_i}(\alpha_i^+) \rightarrow \eta_i^-(u_{i+1})) \ (i = 1, 2, \dots, n-1).$
- 2) If \rightarrow is a *t*-norm implication, then $\tilde{i}_{u_n}(\gamma_n^-) \triangleleft_n$ $(\tilde{i}_{u_n}(\alpha_n^-) \rightarrow \eta_n^-(u_{n+1})) \leq D_n(u_{n+1}) \leq \tilde{s}_{u_n}(\gamma_n^+) \triangleleft_n$ $(\tilde{s}_{u_n}(\alpha_n^+) \rightarrow \eta_n^+(u_{n+1}))$, in which $\gamma_{i+1}^+(u_{i+1}) \triangleq$ $\tilde{s}_{u_i}(\gamma_i^+) \triangleleft_i (\tilde{s}_{u_i}(\alpha_i^+) \rightarrow \eta_i^+(u_{i+1}))$ and $\gamma_{i+1}^-(u_{i+1}) \triangleq$ $\tilde{i}_{u_i}(\gamma_i^-) \triangleleft_i (\tilde{i}_{u_i}(\alpha_i^-) \rightarrow \eta_i^-(u_{i+1}))$ (i = 1, 2, ..., n-1).
- 3) If \rightarrow is an (S, N)-implication $I_{\oplus_1,N}$, then $\tilde{i}_{u_n}\{N(\tilde{s}_{u_n}(\gamma_n^+)) \oplus_1 (N(\tilde{s}_{u_n}(\alpha_n^+)) \oplus_1 \eta_n^-(u_{n+1}))\} \leq D_n(u_{n+1}) \leq \tilde{i}_{u_n}\{N(\tilde{i}_{u_n}(\gamma_n^-)) \oplus_1 (N(\tilde{i}_{u_n}(\alpha_n^-)) \oplus_1 \eta_n^+(u_{n+1}))\}, \text{ in which } \gamma_{i+1}^+(u_{i+1}) \triangleq \tilde{i}_{u_i}\{N(\tilde{i}_{u_i}(\gamma_i^-)) \oplus_1 \eta_i^+(u_{i+1}))\}, \gamma_{i+1}^-(u_{i+1}) \triangleq \tilde{i}_{u_i}\{N(\tilde{s}_{u_i}(\gamma_i^+)) \oplus_1 (N(\tilde{s}_{u_i}(\alpha_i^+)) \oplus_1 \eta_i^-(u_{i+1}))\}, (i = 1, 2, ..., n - 1).$
- 4) If \rightarrow is an QL-implication then $I_{\otimes_1,\oplus_1,N},$ $i_{u_n}\{N(\tilde{s}_{u_n}(\gamma_n^+)) \oplus_1 [i_{u_n}(\gamma_n^-) \otimes_1]$ $(N(\tilde{s}_{u_n}(\alpha_n^+)))$ \oplus_1 \leq $(i_{u_n}(\alpha_n^-) \otimes_1 \eta_n^-(u_{n+1})))]\}$ $D_n(u_{n+1})$ \leq $i_{u_n}\{N(i_{u_n}(\gamma_n^-)) \oplus_1 [\tilde{s}_{u_n}(\gamma_n^+) \otimes_1 (N(i_{u_n}(\alpha_n^-)))]$ ⊕ı $(\tilde{s}_{u_n}(\alpha_n^+) \otimes_1 \eta_n^+(u_{n+1})))]\},$ in which $\gamma_{i+1}^+(u_{i+1})$ ≙ $i_{u_i}\{N(i_{u_i}(\gamma_i^-))\oplus_1[\tilde{s}_{u_i}(\gamma_i^+)\otimes_1(N(i_{u_i}(\alpha_i^-))\oplus_1(\tilde{s}_{u_i}(\alpha_i^+)\otimes_1$ $\eta_i^+(u_{i+1})))]\}, \quad \gamma_{i+1}^-(u_{i+1}) \triangleq \tilde{i}_{u_i}\{N(\tilde{s}_{u_i}(\gamma_i^+)) \oplus_1\}$ $[\tilde{i}_{u_i}(\gamma_i^-) \otimes_1 (N(\tilde{s}_{u_i}(\alpha_i^+)) \oplus_1 (\tilde{i}_{u_i}(\alpha_i^-) \otimes_1 \eta_i^-(u_{i+1})))]\}$ $(i = 1, 2, \ldots, n - 1).$

In accordance with Theorems 5 and 6, if the continuous condition holds for the related operations (i.e., *t*-norm, *t*-conorm, and fuzzy negation), then the fuzzy chain reasoning strategy of BKS is stable, in which \otimes and an (*S*, *N*)-implication, or a QL-implication, or a *t*-norm implication are utilized. In addition, if the *R*-implication and *t*-norm are continuous, then the BKS chain reasoning is stable for interval perturbation.

Suppose that there is a perturbation sequence $([\alpha_{im}^{-}, \alpha_{im}^{+}], [\eta_{im}^{-}, \eta_{im}^{+}], [\gamma_{1m}^{-}, \gamma_{1m}^{+}])$ of the input $(A_1, \ldots, A_n, B_1, \ldots, B_n, C)$ for BKS chain reasoning letting $\lim_{m \to \infty} \sup_{u \in U} (\alpha_{im}^+(u) - \alpha_{im}^-(u)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-(v)) = \lim_{m \to \infty} \sup_{v \in V} (\eta_{im}^+(v) - \alpha_{im}^-($ $\eta_{im}^-(v)$ = $\lim_{m\to\infty} \sup_{u\in U}(\gamma_{1m}^+(u) - \gamma_{1m}^-(u)) = 0$ hold (i = 1, 2, ..., n). In Theorems 5 and 6, we employ λ_m^-, λ_m^+ to represent the corresponding lower and upper bounds of BKS chain output, viz., $D_n(u_{n+1}) \in [\lambda_m^-, \lambda_m^+]$ for $([\alpha_{im}^{-}, \alpha_{im}^{+}], [\eta_{im}^{-}, \eta_{im}^{+}], [\gamma_{1m}^{-}, \gamma_{1m}^{+}]) \quad (m = 1, 2, \ldots).$ When the continuous condition holds, Theorems 5 and 6 imply $\lim_{m\to\infty} \sup_{u_{n+1}\in U_{n+1}} (\lambda_m^+(u_{n+1}) - \lambda_m^-(u_{n+1})) = 0.$ That is, the output of BKS chain reasoning method is converged gradually to a value when the continuous condition is effective. From Definition 15, the BKS chain reasoning method is stable from the perspective of interval perturbation.

V. SIMPLE PERTURBATIONS FOR BKS

A. Simple Perturbations for BKS With One Rule

Here, ten commonly used fuzzy implications are studied.

- 1) $I_1(p,q) = (1-p) \lor q$ (the Kleene-Dienes implication, as an (S, N)-implication and QL-implication).
- 2) $I_2(p,q) = 1 p + pq$ (the Reichenbach implication, as an (*S*, *N*)-implication and QL-implication).
- 3) $I_3(p,q) = 1 \land (1 p + q)$ (the Lukasiewicz implication, as an *R*-implication, (*S*, *N*)-implication and QL-implication).
- 4) $I_4(p,q) = (1-p) \lor (p \land q)$ (the Zadeh implication, as an QL-implication).
- 5) $I_5(p,q) = (p \land q) \lor ((1-p) \land q) \lor ((1-p) \land (1-q))$ ([35]).
- 6) $I_6(p,q) = (1-p) \lor (p+q-1)$ (as an QL-implication [32]).
- 7) $I_7(p,q) = (1-p^2) \lor q$ (as an (*S*, *N*)-implication [36]).
- 8) $I_8(p,q) = 1 p + p^2 q$ (as an QL-implication [32]).
- 9) $I_9(p,q) = p \land q$ (the Mamdani implication, as a *t*-norm implication).
- 10) $I_{10}(p,q) = p \times q$ (the Larsen implication, as a *t*-norm implication).

Lemma 3 [29]: Assume that $A \in F(U), B \in F(V)$, and that β_1 and β_2 are factors of perturbation of A and B. Denote $\Delta I(u, v) = [(A(u) + \beta_1(u)) \rightarrow (B(v) + \beta_2(v))] - [A(u) \rightarrow B(v)]$. Then, one has the following outcomes.

- 1) If \rightarrow employs I_1 , then $(-\beta_1(u)) \land \beta_2(v) \le \Delta I(u, v) \le (-\beta_1(u)) \lor \beta_2(v)$.
- 2) If \rightarrow employs I_2 , then $(-1) \lor (0 \land \beta_2(v) 0 \lor \beta_1(u)) \le \Delta I(u, v) \le 1 \land (0 \lor \beta_2(v) 0 \land \beta_1(u)).$
- 3) If \rightarrow employs I_3 , then $(-1) \lor (0 \land (\beta_2(v) \beta_1(u))) \le \Delta I(u, v) \le 1 \land (0 \lor (\beta_2(v) \beta_1(u))).$
- 4) If \rightarrow employs I_4 , then $(-\beta_1(u)) \land \beta_1(u) \land \beta_2(v) \le \Delta I(u, v) \le (-\beta_1(u)) \lor \beta_1(u) \lor \beta_2(v)$.
- 5) If \rightarrow employs I_5 , then $\beta_1(u) \land \beta_2(v) \land (-\beta_1(u) \lor \beta_2(v)) \le \Delta I(u, v) \le \beta_1(u) \lor \beta_2(v) \lor (-\beta_1(u) \land \beta_2(v)).$
- 6) If \rightarrow employs I_9 , then $\beta_1(u) \land \beta_2(v) \le \Delta I(u, v) \le \beta_1(u) \lor \beta_2(v)$.

We denote $\Delta(v) = BKS(A^*, B^*, C^*)(v) - BKS(A, B, C)(v)$. Thereinto, $A^*(u) = A(u) + \beta_1(u)$, $B^*(v) = B(v) + \beta_2(v)$, and $C^*(u) = C(u) + \beta_3(u)$, in which A^* and C^* are respectively, the perturbations of *A* and *C* on *U* while B^* is the perturbation of *B* on *V*.

In the beginning, we investigate the situation of the BKS-I algorithm whose solution is indicated by (9). It is effortless to give evidence of Lemma 4.

Lemma 4: Suppose that $A \in F(U)$ and $B \in F(V)$, and that β_1 and β_2 are the factors of perturbation of A, B in turn. Then, following results hold.

- 1) If \rightarrow employs I_6 , then $(-1) \lor [(-\beta_1(u)) \land (\beta_1(u) + \beta_2(v))] \le \Delta I(u, v) \le 1 \land [(-\beta_1(u)) \lor (\beta_1(u) + \beta_2(v))].$
- 2) If \to employs I_7 , then $(-1) \lor [(-2A(u)\beta_1(u) (\beta_1(u))^2) \land \beta_2(v)] \le \Delta I(u, v) \le 1 \land [(-2A(u)\beta_1(u) (\beta_1(u))^2) \lor \beta_2(v)].$

 TABLE I

 Estimation of Simple Perturbations for Single Rule Under BKS-I (for Theorem 7)

\rightarrow	Lower bound of $\Delta(v)$	Upper bound of $\Delta(v)$
I_1	$(- ilde{s}_u(eta_1))\wedgeeta_2(v)\wedge(- ilde{s}_u(eta_3))$	$(-\tilde{i}_u(\beta_1)) \lor \beta_2(v) \lor (-\tilde{i}_u(\beta_3))$
I_2	$(-1) \vee [0 \wedge eta_2(v) - (0 \vee ilde{s}_u(eta_1)) - (0 \vee ilde{s}_u(eta_3))]$	$1 \wedge [0 \vee \beta_2(v) - (0 \wedge \tilde{i}_u(\beta_1)) - (0 \wedge \tilde{i}_u(\beta_3))]$
I_3	$(-1) \vee [0 \land (\beta_2(v) - \tilde{s}_u(\beta_1)) + 0 \land (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee (\beta_2(v) - \tilde{i}_u(\beta_1)) + 0 \vee (-\tilde{i}_u(\beta_3))]$
I_4	$(- ilde{s}_u(eta_1))\wedge ilde{i}_u(eta_1)\wedge eta_2(v)\wedge (- ilde{s}_u(eta_3))\wedge ilde{i}_u(eta_3)$	$(-\tilde{i}_u(\beta_1)) \lor \tilde{s}_u(\beta_1) \lor \beta_2(v) \lor (-\tilde{i}_u(\beta_3)) \lor \tilde{s}_u(\beta_3)$
I_5	$(-\tilde{s}_u(eta_1)) \wedge \tilde{i}_u(eta_1) \wedge (-eta_2(v)) \wedge eta_2(v) \wedge (-\tilde{s}_u(eta_3)) \wedge \tilde{i}_u(eta_3)$	$(-\tilde{i}_u(\beta_1)) \vee \tilde{s}_u(\beta_1) \vee (-\beta_2(v)) \vee \beta_2(v) \vee (-\tilde{i}_u(\beta_3)) \vee \tilde{s}_u(\beta_3)$
I_6	$(-1) \vee [(-\tilde{s}_u(\beta_3)) \wedge (\tilde{i}_u(\beta_3) + (-\tilde{s}_u(\beta_1)) \wedge (\tilde{i}_u(\beta_1) + \beta_2(v)))]$	$1 \wedge [(-\tilde{i}_u(\beta_3)) \vee (\tilde{s}_u(\beta_3) + (-\tilde{i}_u(\beta_1)) \vee (\tilde{s}_u(\beta_1) + \beta_2(v)))]$
I	$(-1) \vee [(-0 \vee 2\tilde{s}_u(\beta_1) - \tilde{s}_u(\beta_1^2)) \wedge \beta_2(v) \wedge (-0 \vee 2\tilde{s}_u(\beta_3) -$	$1 \wedge [(-0 \wedge 2\tilde{i}_u(\beta_1) - \tilde{i}_u(\beta_1^2)) \vee \beta_2(v) \vee (0 \wedge 2\tilde{i}_u(\beta_3) -$
17	$- ilde{s}_u(eta_3^2))]$	$\widetilde{i}_u(eta_3^2))]$
I_{\odot}	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) \wedge (-\tilde{s}_u(\beta_1)) + 0 \wedge \beta_2(v) + 0 \wedge \tilde{i}_u(\beta_3) \wedge$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) \vee (-\tilde{i}_u(\beta_1)) + 0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3) \vee$
18	$(- ilde{s}_u(eta_3))]$	$(- ilde{i}_u(eta_3))]$
I_9	$ ilde{i}_u(eta_1)\wedgeeta_2(v)\wedge ilde{i}_u(eta_3)$	$ ilde{s}_u(eta_1) \lor eta_2(v) \lor ilde{s}_u(eta_3)$
I_{10}	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v) + 0 \wedge \tilde{i}_u(\beta_3)]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3)]$

- 3) If \rightarrow employs I_8 , then $\Delta I(u, v) = (A(u)B(v) + A^*(u)B(v) 1) \times \beta_1(u) + (A^*(u))^2\beta_2(v)$, and $(-1) \vee [0 \land \beta_1(u) \land (-\beta_1(u)) + 0 \land \beta_2(v)] \leq \Delta I(u, v) \leq 1 \land [0 \lor \beta_1(u) \lor (-\beta_1(u)) + 0 \lor \beta_2(v)].$
- 4) If \rightarrow employs I_{10} , then $\Delta I(u, v) = B^*(v)\beta_1(u) + A(u)\beta_2(v)$, and $(-1) \lor [0 \land \beta_1(u) + 0 \land \beta_2(v)] \le \Delta I(u, v) \le 1 \land [0 \lor \beta_1(u) + 0 \lor \beta_2(v)].$

We denote $\delta(u, v) = [C(u) + \beta_3(u)] \rightarrow [(A(u) + \beta_1(u)) \rightarrow (B(v) + \beta_2(v))] - [C(u) \rightarrow (A(u) \rightarrow B(v))].$

Theorem 7: Let $A, C \in F(U)$ and $B \in F(V)$, and β_1, β_2 , and β_3 be the factors of perturbation of A, B, and C in turn. If BKS-I is adopted, then one has the outcomes indicated in Table I.

Proof: At present, we exhibit the proof for the situation of I_8 , the others can be gained in an analogous manner. We have from Lemma 4 that

$$\begin{split} \delta(u, v) &= [C(u) + \beta_3(u)] \rightarrow [(A(u) + \beta_1(u)) \rightarrow (B(v) + \beta_2(v))] \\ &- [C(u) \rightarrow (A(u) \rightarrow B(v))] \\ &= 1 - [C(u) + \beta_3(u)] + [C(u) + \beta_3(u)]^2 \\ &\times \left[1 - (A(u) + \beta_1(u)) + (A(u) + \beta_1(u))^2 \\ &\times (B(v) + \beta_2(v))\right] \\ &- \left[1 - C(u) + (C(u))^2 \times \left(1 - A(u) + (A(u))^2 B(v)\right)\right] \\ &= (-\beta_3(u)) + \left[(C(u))^2 + 2C(u)\beta_3(u) + (\beta_3(u))^2\right] \\ &\times [1 - (A(u) + \beta_1(u)) \\ &+ \left((A(u))^2 + (\beta_1(u))^2 + 2A(u)\beta_1(u)\right) \\ &\times (B(v) + \beta_2(v))] \\ &- (C(u))^2 \times \left(1 - A(u) + (A(u))^2 B(v)\right) \\ &= (C(u))^2 \times \Delta I(u, v) + \left[2C(u)\beta_3(u) + (\beta_3(u))^2\right] \\ &\times \left(A^*(u) \rightarrow B^*(v)\right) - \beta_3(u) \\ &= (C(u))^2 \times \left[\left(A(u)B(v) + A^*(u)B(v) - 1\right) \times \beta_1(u) \\ &+ \left(A^*(u)\right)^2 \beta_2(v)\right] \end{split}$$

$$+ [C(u)\beta_{3}(u) + (C(u) + \beta_{3}(u)) \times \beta_{3}(u)] \\ \times (A^{*}(u) \to B^{*}(v)) - \beta_{3}(u) \\ = (C(u))^{2} \times [A(u)B(v) + A^{*}(u)B(v) - 1] \times \beta_{1}(u) \\ + (A^{*}(u))^{2}(C(u))^{2}\beta_{2}(v) \\ + [C(u) \times (A^{*}(u) \to B^{*}(v)) \\ + C^{*}(u) \times (A^{*}(u) \to B^{*}(v)) - 1] \times \beta_{3}(u).$$

On the strength of Lemmas 1 and 2, we acquire

$$\begin{split} \Delta(v) &= \inf_{u \in U} \{ [C(u) + \beta_3(u)] \\ &\to [(A(u) + \beta_1(u)) \to (B(v) + \beta_2(v))] \} \\ &- \inf_{u \in U} \{ [C(u) \to (A(u) \to B(v))] \} \\ &\leq \sup_{u \in U} \{ [C(u) + \beta_3(u)] \\ &\to [(A(u) + \beta_1(u)) \to (B(v) + \beta_2(v))] \\ &- [C(u) \to (A(u) \to B(v))] \} \\ &\leq \sup_{u \in U} \{ (C(u))^2 \times [A(u)B(v) + A^*(u)B(v) - 1] \times \beta_1(u) \\ &+ (A^*(u))^2 (C(u))^2 \beta_2(v) \\ &+ [C(u) \times (A^*(u) \to B^*(v)) \\ &+ C^*(u) \times (A^*(u) \to B^*(v)) - 1] \times \beta_3(u) \} \\ &\leq \sup_{u \in U} \{ (C(u))^2 \times [A(u)B(v) + A^*(u)B(v) - 1] \times \beta_1(u) \} \\ &+ \sup_{u \in U} \{ (A^*(u))^2 (C(u))^2 \beta_2(v) \} \\ &+ \sup_{u \in U} \{ (A^*(u))^2 (C(u))^2 \beta_2(v) \} \\ &+ \sup_{u \in U} \{ [C(u) \times (A^*(u) \to B^*(v)) + C^*(u) \\ &\times (A^*(u) \to B^*(v)) - 1] \times \beta_3(u) \}. \end{split}$$

In this formula, we determine $\sup_{u \in U} \{ (C(u))^2 \times [A(u)B(v) + A^*(u)B(v) - 1] \times \beta_1(u) \}.$

- 1) If $\sup_{u \in U} \{ (C(u))^2 \times [A(u)B(v) + A^*(u)B(v) 1] \times \beta_1(u) \} \le 0$, then the outcome is evident.
- 2) If $\sup_{u \in U} \{ (C(u))^2 \times [A(u)B(v) + A^*(u)B(v) 1] \times \beta_1(u) \} > 0$, then two situations are considered. On the one hand, if $\sup_{u \in U} \{ \beta_1(u) \} > 0$, then $\sup_{u \in U} \{ (C(u))^2 \times (C(u))^2 \times (C(u))^2 + (C(u))^2$

 TABLE II

 Estimation of Simple Perturbations for Single Rule Under BKS-T With \otimes_M (for Theorem 8)

\rightarrow	Lower bound of $\Delta(v)$	Upper bound of $\Delta(v)$
I_1	$ ilde{i}_u(eta_1)\wedgeeta_2(v)\wedge(- ilde{s}_u(eta_3))$	$\tilde{s}_u(eta_1) \lor eta_2(v) \lor (-\tilde{i}_u(eta_3))$
L	$(-1) \vee [(0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \tilde{i}_u(\beta_3)) \wedge (0 \wedge \beta_2(v) + 0 \wedge \tilde{i}_u(\beta_3))$	$1 \wedge [(0 \vee \tilde{s}_u(\beta_1) + 0 \vee \tilde{s}_u(\beta_3)) \vee (0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3))$
12	$- ilde{s}_u(eta_3)]$	$- ilde{i}_u(eta_3)]$
I_3	$(-1) \vee [0 \wedge ((\tilde{i}_u(\beta_1) \wedge \beta_2(v)) - \tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee ((\tilde{s}_u(\beta_1) \vee \beta_2(v)) - \tilde{i}_u(\beta_3))]$
I_4	$ ilde{i}_u(eta_1) \wedge eta_2(v) \wedge (- ilde{s}_u(eta_3)) \wedge ilde{i}_u(eta_3)$	$ ilde{s}_u(eta_1) \lor eta_2(v) \lor (- ilde{i}_u(eta_3)) \lor ilde{s}_u(eta_3)$
I_5	$(-\tilde{s}_u(\beta_1)) \wedge \tilde{i}_u(\beta_1) \wedge (-\beta_2(v)) \wedge \beta_2(v) \wedge (-\tilde{s}_u(\beta_3)) \wedge \tilde{i}_u(\beta_3)$	$(-\tilde{i}_u(\beta_1)) \vee \tilde{s}_u(\beta_1) \vee (-\beta_2(v)) \vee \beta_2(v) \vee (-\tilde{i}_u(\beta_3)) \vee \tilde{s}_u(\beta_3)$
I_6	$(-1) \lor [(-\tilde{s}_u(\beta_3)) \land (\tilde{i}_u(\beta_3) + \tilde{i}_u(\beta_1) \land \beta_2(v))]$	$1 \wedge [(-\tilde{i}_u(\beta_3)) \vee (\tilde{s}_u(\beta_3) + \tilde{s}_u(\beta_1) \vee \beta_2(v))]$
I_7	$(-1) \vee [\tilde{i}_u(\beta_1) \land \beta_2(v) \land (-0 \lor 2\tilde{s}_u(\beta_3) - \tilde{s}_u(\beta_3^2))]$	$1 \wedge [\tilde{s}_u(\beta_1) \vee \beta_2(v) \vee (-0 \wedge 2\tilde{i}_u(\beta_3) - \tilde{i}_u(\beta_3^2))]$
L	$(-1) \vee [(0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \tilde{i}_u(2\beta_3) + 0 \wedge \tilde{i}_u(\beta_3^2)) \wedge$	$1 \wedge [(0 \vee \tilde{s}_u(\beta_1) + 0 \vee \tilde{s}_u(2\beta_3) + 0 \vee \tilde{s}_u(\beta_3^2)) \vee$
18	$(0 \wedge eta_2(v) + 0 \wedge \widetilde{i}_u(2eta_3) + 0 \wedge \widetilde{i}_u(eta_3^2)) - \widetilde{s}_u(eta_3)]$	$(0 \lor \beta_2(v) + 0 \lor \tilde{s}_u(2\beta_3) + 0 \lor \tilde{s}_u(\beta_3^2)) - \tilde{i}_u(\beta_3)]$
I_9	$ ilde{i}_u(eta_1)\wedgeeta_2(v)\wedge ilde{i}_u(eta_3)$	$ ilde{s}_u(eta_1) ee eta_2(v) ee ilde{s}_u(eta_3)$
I_{10}	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \tilde{i}_u(\beta_3)] \wedge [0 \wedge \beta_2(v) + 0 \wedge \tilde{i}_u(\beta_3)]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \tilde{s}_u(\beta_3)] \vee [0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3)]$

 TABLE III

 Estimation of Simple Perturbations for Single Rule Under BKS-T With \otimes_P (for Theorem 9)

\rightarrow	Lower bound of $\Delta(v)$	Upper bound of $\Delta(v)$
I_1	$(-1) \vee [(0 \wedge \tilde{i}_u(eta_1) + 0 \wedge eta_2(v)) \wedge (-\tilde{s}_u(eta_3))]$	$1 \wedge \left[\left(0 \lor \tilde{s}_u(\beta_1) + 0 \lor \beta_2(v) \right) \lor \left(-\tilde{i}_u(\beta_3) \right) \right]$
I_2	$(-1) \vee [0 \wedge \tilde{i}_u(eta_1) + 0 \wedge eta_2(v) - (0 \vee \tilde{s}_u(eta_3))]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) - (0 \wedge \tilde{i}_u(\beta_3))]$
I_3	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v) + 0 \wedge (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) + 0 \vee (-\tilde{i}_u(\beta_3))]$
I_4	$(-1) \lor [(0 \land \tilde{i}_u(eta_1) + 0 \land eta_2(v)) \land (-\tilde{s}_u(eta_3)) \land \tilde{i}_u(eta_3)]$	$1 \wedge [(0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v)) \vee (-\tilde{i}_u(\beta_3)) \vee \tilde{s}_u(\beta_3)]$
Ι	$(-1) \vee [(-0 \vee \tilde{s}_u(\beta_1) - 0 \vee \beta_2(v)) \land (0 \land \tilde{i}_u(\beta_1) + 0 \land \beta_2(v)) \land$	$1 \wedge [(-0 \wedge \tilde{i}_u(\beta_1) - 0 \wedge \beta_2(v)) \vee (0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v)) \vee$
15	$(- ilde{s}_u(eta_3))\wedge ilde{i}_u(eta_3)]$	$(-\widetilde{i}_u(eta_3)) ee \widetilde{s}_u(eta_3)]$
I_6	$(-1) \lor [(-\tilde{s}_u(\beta_3)) \land (0 \land \tilde{i}_u(\beta_1) + 0 \land \beta_2(v) + \tilde{i}_u(\beta_3))]$	$1 \wedge [(-\tilde{i}_u(\beta_3)) \vee (0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) + \tilde{s}_u(\beta_3))]$
I_7	$(-1) \vee [(0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v)) \wedge (-0 \vee 2\tilde{s}_u(\beta_3) - \tilde{s}_u(\beta_3^2))]$	$1 \wedge [(0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v)) \vee (-0 \wedge 2\tilde{i}_u(\beta_3) - \tilde{i}_u(\beta_3^2))]$
I_8	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v) + 0 \wedge \tilde{i}_u(\beta_3) \wedge (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3) \vee (-\tilde{i}_u(\beta_3))]$
I_9	$(-1) \vee [(0 \wedge \tilde{i}_u(eta_1) + 0 \wedge eta_2(v)) \wedge \tilde{i}_u(eta_3)]$	$1 \wedge [(0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v)) \vee \tilde{s}_u(\beta_3)]$
I_{10}	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v) + 0 \wedge \tilde{i}_u(\beta_3)]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3)]$

 $[A(u)B(v) + A^{*}(u)B(v) - 1] \times \beta_{1}(u)\} \leq \sup_{u \in U} \{\beta_{1}(u)\} = \tilde{s}_{u}(\beta_{1}). \text{ On the other hand, if } \sup_{u \in U} \{\beta_{1}(u)\} < 0, \text{ then } \sup_{u \in U} \{(C(u))^{2} \times [A(u)B(v) + A^{*}(u)B(v) - 1] \times \beta_{1}(u)\} \leq \sup_{u \in U} \{(-\beta_{1}(u))\} = -\inf_{u \in U} \{\beta_{1}(u)\} = -\tilde{i}_{u}(\beta_{1}).$

Placing these two expressions together, we gain $\sup_{u \in U} \{ (C(u))^2 \times [A(u)B(v) + A^*(u)B(v) - 1] \times \beta_1(u) \} \le 0 \lor \tilde{s}_u(\beta_1) \lor (-\tilde{i}_u(\beta_1)).$

Analogously, we obtain $\sup_{u \in U} \{ (A^*(u))^2 (C(u))^2 \beta_2(v) \} \le 0 \lor \beta_2(v), \text{ and } \sup_{u \in U} \{ [C(u) \times (A^*(u) \to B^*(v)) + C^*(u) \times (A^*(u) \to B^*(v)) - 1] \times \beta_3(u) \} \le 0 \lor \tilde{s}_u(\beta_3) \lor (-\tilde{i}_u(\beta_3)).$

As a consequence, we have $\Delta(v) \leq 0 \vee \tilde{s}_u(\beta_1) \vee (-\tilde{i}_u(\beta_1)) + 0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3) \vee (-\tilde{i}_u(\beta_3)).$

Distinctly, $\Delta(v) \leq 1$, hence it implies $\Delta(v) \leq 1 \wedge [0 \vee \tilde{s}_u(\beta_1) \vee (-\tilde{i}_u(\beta_1)) + 0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3) \vee (-\tilde{i}_u(\beta_3))].$

In an analogous situation, we obtain $\Delta(v) \ge (-1) \lor [0 \land \tilde{i}_u(\beta_1) \land (-\tilde{s}_u(\beta_1)) + 0 \land \beta_2(v) + 0 \land \tilde{i}_u(\beta_3) \land (-\tilde{s}_u(\beta_3))].$

Summarizing the above findings, one achieves $(-1) \vee [0 \land \tilde{i}_u(\beta_1) \land (-\tilde{s}_u(\beta_1)) + 0 \land \beta_2(v) + 0 \land \tilde{i}_u(\beta_3) \land (-\tilde{s}_u(\beta_3))] \le \Delta(v) \le 1 \land [0 \lor \tilde{s}_u(\beta_1) \lor (-\tilde{i}_u(\beta_1)) + 0 \lor \beta_2(v) + 0 \lor \tilde{s}_u(\beta_3) \lor (-\tilde{i}_u(\beta_3))].$

In addition, we reveal the situation of the BKS-T algorithm, that is, the solution is demonstrated by (8). It is easy to gain Lemma 5.

Lemma 5: Suppose that $A \in F(U)$ and $B \in F(V)$, and that β_1 and β_2 are factors of perturbation of A and B in turn. Denote $\Delta T(u, v) = [(A(u) + \beta_1(u)) \otimes (B(v) + \beta_2(v))] - [A(u) \otimes B(v)]$. Then, we have the following outcomes.

- 1) If \otimes is \otimes_M , then $\beta_1(u) \wedge \beta_2(v) \leq \Delta T(u, v) \leq \beta_1(u) \vee \beta_2(v)$.
- 2) If \otimes employs \otimes_P , then $\Delta T(u, v) = B^*(v)\beta_1(u) + A(u)\beta_2(v)$, and $(-1) \vee [0 \wedge \beta_1(u) + 0 \wedge \beta_2(v)] \leq \Delta T(u, v) \leq 1 \wedge [0 \vee \beta_1(u) + 0 \vee \beta_2(v)].$
- 3) If \otimes takes \otimes_L , then $(-1) \vee [(\beta_1(u) + \beta_2(v)) \wedge 0] \leq \Delta T(u, v) \leq 1 \wedge [(\beta_1(u) + \beta_2(v)) \vee 0].$

Theorems 8-10 are validated as the previous ones.

Theorem 8: Suppose that $A, C \in F(U)$ and $B \in F(V)$, and β_1, β_2 , and β_2 are the factors of perturbation of A, B, and C in turn. If BKS-T is employed and \otimes is \otimes_M , then the results in Table II hold for I_1, I_2, \ldots, I_{10} .

Theorem 9: Suppose that $A, C \in F(U)$ and $B \in F(V)$, and that β_1, β_2 , and β_2 are the factors of perturbation of A, B, and C in turn. If BKS-T is utilized, and \otimes takes \otimes_P , then we obtain the outcomes in Table III for I_1 - I_{10} .

Theorem 10: Suppose that $A, C \in F(U)$ and $B \in F(V)$, and that β_1, β_2 , and β_2 are the factors of perturbation of A, B,

TABLE IV Estimation of Simple Perturbations for Single Rule Under BKS-T With \otimes_L (for Theorem 10)

\rightarrow	Lower bound of $\Delta(v)$	Upper bound of $\Delta(v)$
I_1	$(-1) \vee [0 \land (\tilde{i}_u(\beta_1) + \beta_2(v)) \land (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee (\tilde{s}_u(\beta_1) + \beta_2(v)) \vee (-\tilde{i}_u(\beta_3))]$
I_2	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v) - 0 \vee \tilde{s}_u(\beta_3)]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) - 0 \wedge \tilde{i}_u(\beta_3)]$
I_3	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v) + 0 \wedge (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) + 0 \vee (-\tilde{i}_u(\beta_3))]$
I_4	$(-1) \vee [(0 \land (\tilde{i}_u(\beta_1) + \beta_2(v))) \land (-\tilde{s}_u(\beta_3)) \land \tilde{i}_u(\beta_3)]$	$1 \wedge [(0 \vee (\tilde{s}_u(\beta_1) + \beta_2(v))) \vee (-\tilde{i}_u(\beta_3)) \vee \tilde{s}_u(\beta_3)]$
<i>I</i> -	$(-1) \vee [0 \wedge (-\tilde{s}_u(\beta_1) - \beta_2(v)) \wedge (\tilde{i}_u(\beta_1) + \beta_2(v)) \wedge (-\tilde{s}_u(\beta_3))$	$1 \wedge [0 \vee (-\tilde{i}_u(\beta_1) - \beta_2(v)) \vee (\tilde{s}_u(\beta_1) + \beta_2(v)) \vee (-\tilde{i}_u(\beta_3))$
15	$\wedge \tilde{i}_u(\beta_3)]$	$ee ilde s_u(eta_3)]$
I_6	$(-1) \vee [(-\tilde{s}_u(\beta_3)) \wedge (0 \wedge (\tilde{i}_u(\beta_1) + \beta_2(v)) + \tilde{i}_u(\beta_3))]$	$1 \wedge [(-\tilde{i}_u(\beta_3)) \vee (0 \vee (\tilde{s}_u(\beta_1) + \beta_2(v)) + \tilde{s}_u(\beta_3))]$
I_7	$(-1) \vee [0 \wedge (\tilde{i}_u(\beta_1) + \beta_2(v)) \wedge (-0 \vee 2\tilde{s}_u(\beta_3) - \tilde{s}_u(\beta_3^2))]$	$1 \wedge [0 \vee (\tilde{s}_u(\beta_1) + \beta_2(v)) \vee (-0 \wedge 2\tilde{i}_u(\beta_3) - \tilde{i}_u(\beta_3^2))]$
I_8	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v) + 0 \wedge \tilde{i}_u(\beta_3) \wedge (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3) \vee (-\tilde{i}_u(\beta_3))]$
I_9	$(-1) \vee [0 \land (\tilde{i}_u(\beta_1) + \beta_2(v)) \land \tilde{i}_u(\beta_3)]$	$1 \wedge [0 \vee (\tilde{s}_u(\beta_1) + \beta_2(v)) \vee \tilde{s}_u(\beta_3)]$
I_{10}	$(-1) \vee [0 \wedge \tilde{i}_u(\beta_1) + 0 \wedge \beta_2(v) + 0 \wedge \tilde{i}_u(\beta_3) \wedge (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee \tilde{s}_u(\beta_1) + 0 \vee \beta_2(v) + 0 \vee \tilde{s}_u(\beta_3) \vee (-\tilde{i}_u(\beta_3))]$

TABLE V	
ESTIMATION OF SIMPLE PERTURBATIONS FOR MULTIPLE RULES UNDER BKS-I (FOR THEOREM 1	11)

	Leven being Lef A ()	Under head of $A_{(a)}$
\rightarrow	Lower bound of $\Delta_n(v)$	Upper bound of $\Delta_n(v)$
I_1	$[\wedge_{i=1}^{n}(-\tilde{s}_{u}(\beta_{1i}))] \wedge [\wedge_{i=1}^{n}\beta_{2i}(v)] \wedge (-\tilde{s}_{u}(\beta_{3}))$	$[\vee_{i=1}^n(-\tilde{i}_u(\beta_{1i}))] \vee [\vee_{i=1}^n\beta_{2i}(v)] \vee (-\tilde{i}_u(\beta_3))$
I_2	$(-1) \vee [0 \wedge (\wedge_{i=1}^n \beta_{2i}(v)) - 0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) - 0 \vee \tilde{s}_u(\beta_3)]$	$1 \wedge [0 \vee (\vee_{i=1}^n \beta_{2i}(v)) - 0 \wedge (\wedge_{i=1}^n \tilde{i}_u(\beta_{1i})) - 0 \wedge \tilde{i}_u(\beta_3)]$
I_3	$(-1) \vee [0 \wedge \wedge_{i=1}^{n} (\beta_{2i}(v) - \tilde{s}_u(\beta_{1i})) + 0 \wedge (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee \vee_{i=1}^{n} (\beta_{2i}(v) - \tilde{i}_{u}(\beta_{1i})) + 0 \vee (-\tilde{i}_{u}(\beta_{3}))]$
L	$[\wedge_{i=1}^{n}(-\tilde{s}_{u}(\beta_{1i}))] \wedge [\wedge_{i=1}^{n}\tilde{i}_{u}(\beta_{1i})] \wedge [\wedge_{i=1}^{n}\beta_{2i}(v)] \wedge (-\tilde{s}_{u}(\beta_{3}))$	$[\vee_{i=1}^n(-\tilde{i}_u(\beta_{1i}))] \vee [\vee_{i=1}^n \tilde{s}_u(\beta_{1i})] \vee [\vee_{i=1}^n \beta_{2i}(v)] \vee (-\tilde{i}_u(\beta_3))$
14	$\wedge \tilde{i}_u(\beta_3)$	$ee ilde s_u(eta_3)$
I-	$(\wedge_{i=1}^{n}(-\tilde{s}_{u}(\beta_{1i}))) \wedge (\wedge_{i=1}^{n}\tilde{i}_{u}(\beta_{1i})) \wedge (\wedge_{i=1}^{n}(-\beta_{2i}(v))) \wedge$	$(\vee_{i=1}^{n}(-\tilde{i}_{u}(\beta_{1i}))) \vee (\vee_{i=1}^{n}\tilde{s}_{u}(\beta_{1i})) \vee (\vee_{i=1}^{n}(-\beta_{2i}(v))) \vee$
15	$(\wedge_{i=1}^n \beta_{2i}(v)) \wedge (-\tilde{s}_u(\beta_3)) \wedge \tilde{i}_u(\beta_3)$	$(\vee_{i=1}^n \beta_{2i}(v)) \vee (-\tilde{i}_u(\beta_3)) \vee \tilde{s}_u(\beta_3)$
L	$(-1) \vee [(-\tilde{s}_u(\beta_3)) \wedge (\tilde{i}_u(\beta_3) + (\wedge_{i=1}^n (-\tilde{s}_u(\beta_{1i})) \wedge$	$1 \wedge [(-\tilde{i}_u(\beta_3)) \vee (\tilde{s}_u(\beta_3) + (\vee_{i=1}^n (-\tilde{i}_u(\beta_{1i}))) \vee$
10	$(\wedge_{i=1}^{n}\tilde{i}_{u}(\beta_{1i}) + \wedge_{i=1}^{n}\beta_{2i}(v))))]$	$(\vee_{i=1}^{n}\tilde{s}_{u}(\beta_{1i}) + \vee_{i=1}^{n}\beta_{2i}(v)))))$
I_7	$(-1) \vee [(\wedge_{i=1}^n (-0 \vee 2\tilde{s}_u(\beta_{1i}) - \tilde{s}_u(\beta_{1i}^2))) \wedge (\wedge_{i=1}^n \beta_{2i}(v)) \wedge$	$1 \wedge [(\vee_{i=1}^n (-0 \wedge 2\tilde{i}_u(\beta_{1i}) - \tilde{i}_u(\beta_{1i}^2))) \vee (\vee_{i=1}^n \beta_{2i}(v)) \vee$
	$(-0 \lor 2 ilde{s}_u(eta_3) - ilde{s}_u(eta_3^2))]$	$(-0 \wedge 2 ilde{i}_u(eta_3) - ilde{i}_u(eta_3^2))]$
L	$(-1) \vee [0 \wedge (\wedge_{i=1}^n \tilde{i}_u(\beta_1)) \wedge (\wedge_{i=1}^n (-\tilde{s}_u(\beta_1))) +$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_1)) \vee (\vee_{i=1}^n (-\tilde{i}_u(\beta_1))) +$
18	$0 \wedge (\wedge_{i=1}^{n} \beta_{2i}(v)) + 0 \wedge \tilde{i}_{u}(\beta_{3}) \wedge (-\tilde{s}_{u}(\beta_{3}))]$	$0 \lor (\lor_{i=1}^n \beta_{2i}(v)) + 0 \lor \tilde{s}_u(\beta_3) \lor (-\tilde{i}_u(\beta_3))]$
I_9	$(\wedge_{i=1}^{n}\tilde{i}_{u}(\beta_{1i}))\wedge(\wedge_{i=1}^{n}\beta_{2i}(v))\wedge\tilde{i}_{u}(\beta_{3})$	$(\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) \vee (\vee_{i=1}^n \beta_{2i}(v)) \vee \tilde{s}_u(\beta_3)$
I_{10}	$(-1) \vee [0 \land (\land_{i=1}^{n} \tilde{i}_{u}(\beta_{1i})) + 0 \land (\land_{i=1}^{n} \beta_{2i}(v)) + 0 \land \tilde{i}_{u}(\beta_{3})]$	$1 \wedge [0 \vee (\vee_{i=1}^{n} \tilde{s}_{u}(\beta_{1i})) + 0 \vee (\vee_{i=1}^{n} \beta_{2i}(v)) + 0 \vee \tilde{s}_{u}(\beta_{3})]$

and *C* in turn. If BKS-T is used and \otimes takes \otimes_L , then the conclusions in Table IV are derived.

Suppose that $A, C \in F(U)$ and $B \in F(V)$. If there is a small positive real number ε , and functions $\theta_k(u) : U \to [-1, 1]$ for k = 1, 3 or $\theta_2(v) : V \to [-1, 1]$ letting $\beta_k(u) = \varepsilon \theta_k(u)$ hold where k = 1, 3 and $\beta_2(v) = \varepsilon \theta_2(v)$, then we acquire from Theorems 7–10 that $\lim_{\varepsilon \to 0} BKS(A^*, B^*, C^*)(v) = BKS(A, B, C)(v)$. Hence, the BKS algorithm is stable in view of Definition 16-1).

From another perspective, assume that there is a perturbation sequence $\{(A_m^*, B_m^*, C_m^*)\}$ with regard to $\{(\beta_{1m}(u), \beta_{2m}(v), \beta_{3m}(u))\}$, viz., $A_m^*(u) = A(u) + \beta_{1m}(u)$, $B_m^*(v) = B(v) + \beta_{2m}(v)$, and $C_m^*(u) = C(u) + \beta_{3m}(u)$ (m = 1, 2, ...,)). Let $\lim_{m\to\infty} \sup_{u\in U} |\beta_{1m}(u)| = \lim_{m\to\infty} \sup_{v\in V} |\beta_{2m}(v)| = \lim_{m\to\infty} \sup_{u\in U} |\beta_{3m}(u)| = 0$ holds for $\beta_{1m}(u), \beta_{2m}(v), \beta_{3m}(u)$. Then, from Theorems 7–10, it follows that $\lim_{m\to\infty} BKS(A_m^*, B_m^*, C_m^*)(v) = BKS(A, B, C)(v)$. In consequence, it follows from Definition 16-2) that the BKS algorithm is asymptotic stable. All in all, the BKS algorithm for the situation of a single rule is stable from the perspective of simple perturbation.

B. Simple Perturbations for BKS With Multiple Rules

For the situation of multiple rules, we employ the notation $\Delta_n(v) = BKS(A_1^*, \ldots, A_n^*, B_1^*, \ldots, B_n^*, C^*)(v) - BKS(A_1, \ldots, A_n, B_1, \ldots, B_n, C)(v)$. Thereinto, $A_i^*(u) = A_i(u) + \beta_{1i}(u)$, $B_i^*(v) = B_i(v) + \beta_{2i}(v)$, and $C^*(u) = C(u) + \beta_3(u)$, in which A_i^* and C^* are, respectively, the perturbations of A_i , C on U while B_i^* is the perturbation of B_i on V $(i = 1, 2, \ldots, n)$.

First, we analyze the BKS-I algorithm, viz., the expression (7). Denote $\delta_n(u, v) = [(C(u) + \beta_3(u)) \rightarrow \wedge_{i=1}^n ((A_i(u) + \beta_{1i}(u)) \rightarrow (B_i(v) + \beta_{2i}(v)))] - [C(u) \rightarrow \wedge_{i=1}^n (A_i(u) \rightarrow B_i(v))]$. Theorem 11 can be verified in a similar fashion.

Theorem 11: Assume that $A_i, C \in F(U)$ and $B_i \in F(V)$, and that β_{1i}, β_{2i} , and β_3 are the factors of perturbation of A_i, B_i , and C in turn (i = 1, 2, ..., n). If BKS-I is employed, then the results for I_1 - I_{10} are given below in Table V.

6278

TABLE VI	
ESTIMATION OF SIMPLE PERTURBATIONS FOR MULTIPLE RULES	UNDER BKS-T WITH \otimes_M (for Theorem 12)

\rightarrow	Lower bound of $\Delta_n(v)$	Upper bound of $\Delta_n(v)$
I_1	$(\wedge_{i=1}^n \tilde{i}_u(eta_{1i})) \wedge (\wedge_{i=1}^n eta_{2i}(v)) \wedge (-\tilde{s}_u(eta_3))$	$(\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) \vee (\vee_{i=1}^n \beta_{2i}(v)) \vee (-\tilde{i}_u(\beta_3))$
I_2	$(-1) \vee [(0 \land (\land_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \land \tilde{i}_u(\beta_3)) \land (0 \land (\land_{i=1}^n \beta_{2i}(v)) +$	$1 \wedge [(0 \lor (\lor_{i=1}^{n} \tilde{s}_{u}(\beta_{1i})) + 0 \lor \tilde{s}_{u}(\beta_{3})) \lor (0 \lor (\lor_{i=1}^{n} \beta_{2i}(v)) +$
12	$0 \wedge ilde{i}_u(eta_3)) - ilde{s}_u(eta_3)]$	$0 ee ilde{s}_u(eta_3)) - ilde{i}_u(eta_3)]$
I_3	$(-1) \vee [0 \wedge (((\wedge_{i=1}^n \tilde{i}_u(\beta_{1i})) \wedge (\wedge_{i=1}^n \beta_{2i}(v))) - \tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee (((\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) \vee (\vee_{i=1}^n \beta_{2i}(v))) - \tilde{i}_u(\beta_3))]$
I_4	$(\wedge_{i=1}^{n}\tilde{i}_{u}(\beta_{1i}))\wedge(\wedge_{i=1}^{n}\beta_{2i}(v))\wedge(-\tilde{s}_{u}(\beta_{3}))\wedge\tilde{i}_{u}(\beta_{3})$	$(\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) \vee (\vee_{i=1}^n \beta_{2i}(v)) \vee (-\tilde{i}_u(\beta_3)) \vee \tilde{s}_u(\beta_3)$
I۳	$(\wedge_{i=1}^{n}(-\tilde{s}_{u}(\beta_{1i}))) \wedge (\wedge_{i=1}^{n}\tilde{i}_{u}(\beta_{1i})) \wedge (\wedge_{i=1}^{n}(-\beta_{2i}(v))) \wedge$	$(\vee_{i=1}^{n}(-\tilde{i}_{u}(\beta_{1i}))) \vee (\vee_{i=1}^{n}\tilde{s}_{u}(\beta_{1i})) \vee (\vee_{i=1}^{n}(-\beta_{2i}(v))) \vee$
10	$(\wedge_{i=1}^n \beta_{2i}(v)) \wedge (-\tilde{s}_u(\beta_3)) \wedge \tilde{i}_u(\beta_3)$	$(\vee_{i=1}^n\beta_{2i}(v))\vee(-\tilde{i}_u(\beta_3))\vee\tilde{s}_u(\beta_3)$
I_6	$(-1) \lor [(-\tilde{s}_u(\beta_3)) \land (\tilde{i}_u(\beta_3) + (\land_{i=1}^n \tilde{i}_u(\beta_{1i})) \land (\land_{i=1}^n \beta_{2i}(v)))]$	$1 \wedge [(-\tilde{i}_u(\beta_3)) \vee (\tilde{s}_u(\beta_3) + (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) \vee (\vee_{i=1}^n \beta_{2i}(v)))]$
I_7	$(-1) \lor \left[\left(\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i}) \right) \land \left(\wedge_{i=1}^{n} \beta_{2i}(v) \right) \land \left(-0 \lor 2\tilde{s}_{u}(\beta_{3}) - \tilde{s}_{u}(\beta_{3}^{2}) \right) \right]$	$1 \wedge [(\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) \vee (\vee_{i=1}^n \beta_{2i}(v)) \vee (-0 \wedge 2\tilde{i}_u(\beta_3) - \tilde{i}_u(\beta_3^2))]$
I_{\circ}	$(-1) \vee [(0 \wedge \wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i}) + 0 \wedge \tilde{i}_{u}(2\beta_{3}) + 0 \wedge \tilde{i}_{u}(\beta_{3}^{2})) \wedge$	$1 \wedge [(0 \vee \bigvee_{i=1}^n \tilde{s}_u(\beta_{1i}) + 0 \vee \tilde{s}_u(2\beta_3) + 0 \vee \tilde{s}_u(\beta_3^2)) \vee$
18	$(0 \wedge \wedge_{i=1}^{n} \beta_{2i}(v) + 0 \wedge \tilde{i}_{u}(2\beta_{3}) + 0 \wedge \tilde{i}_{u}(\beta_{3}^{2})) - \tilde{s}_{u}(\beta_{3})]$	$(0 \lor \lor_{i=1}^{n} \beta_{2i}(v) + 0 \lor \tilde{s}_u(2\beta_3) + 0 \lor \tilde{s}_u(\beta_3^2)) - \tilde{i}_u(\beta_3)]$
I_9	$(\wedge_{i=1}^{n}\tilde{i}_{u}(\beta_{1i}))\wedge(\wedge_{i=1}^{n}\beta_{2i}(v))\wedge\tilde{i}_{u}(\beta_{3})$	$(\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) \vee (\vee_{i=1}^n \beta_{2i}(v)) \vee \tilde{s}_u(\beta_3)$
I_{10}	$(-1) \vee [(0 \land (\land_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \land \tilde{i}_u(\beta_3)) \land (0 \land (\land_{i=1}^n \beta_{2i}(v)) +$	$1 \wedge [(0 \lor (\lor_{i=1}^{n} \tilde{s}_{u}(\beta_{1i})) + 0 \lor \tilde{s}_{u}(\beta_{3})) \lor (0 \lor (\lor_{i=1}^{n} \beta_{2i}(v)) +$
110	$0\wedge ilde{i}_u(eta_3))$	$0 ee ilde s_u(eta_3))]$

TABLE VII

ESTIMATION OF SIMPLE PERTURBATIONS FOR MULTIPLE RULES UNDER BKS-T WITH \otimes_P (for Theorem 13)

\rightarrow	Lower bound of $\Delta_n(v)$	Upper bound of $\Delta_n(v)$
I_1	$(-1) \vee \left[\left(0 \land \left(\land_{i=1}^{n} \tilde{i}_{u}(\beta_{1i}) \right) + 0 \land \left(\land_{i=1}^{n} \beta_{2i}(v) \right) \right) \land \left(-\tilde{s}_{u}(\beta_{3}) \right) \right]$	$1 \wedge [(0 \lor (\lor_{i=1}^{n} \tilde{s}_{u}(\beta_{1i})) + 0 \lor (\lor_{i=1}^{n} \beta_{2i}(v))) \lor (-\tilde{i}_{u}(\beta_{3}))]$
I_2	$(-1) \vee [0 \wedge (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i})) + 0 \wedge (\wedge_{i=1}^{n} \beta_{2i}(v)) - (0 \vee \tilde{s}_{u}(\beta_{3}))]$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) + 0 \vee (\vee_{i=1}^n \beta_{2i}(v)) - (0 \wedge \tilde{i}_u(\beta_3))]$
I_3	$(-1) \vee [0 \wedge (\wedge_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \wedge (\wedge_{i=1}^n \beta_{2i}(v)) + 0 \wedge (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee (\vee_{i=1}^{n} \tilde{s}_{u}(\beta_{1i})) + 0 \vee (\vee_{i=1}^{n} \beta_{2i}(v)) + 0 \vee (-\tilde{i}_{u}(\beta_{3}))]$
I_4	$(-1) \vee \left[(0 \wedge \bigwedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i}) + 0 \wedge \bigwedge_{i=1}^{n} \beta_{2i}(v)) \wedge (-\tilde{s}_{u}(\beta_{3})) \wedge \tilde{i}_{u}(\beta_{3}) \right]$	$1 \wedge [(0 \vee \bigvee_{i=1}^{n} \tilde{s}_{u}(\beta_{1i}) + 0 \vee \bigvee_{i=1}^{n} \beta_{2i}(v)) \vee (-\tilde{i}_{u}(\beta_{3})) \vee \tilde{s}_{u}(\beta_{3})]$
I_5	$(-1) \lor [\wedge_{i=1}^{n} (-0 \lor \tilde{s}_{u}(\beta_{1i}) - 0 \lor \beta_{2i}(v)) \land (0 \land (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i})) +$	$1 \wedge [\vee_{i=1}^n (-0 \wedge \tilde{i}_u(\beta_{1i}) - 0 \wedge \beta_{2i}(v)) \vee (0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) +$
	$0 \wedge (\wedge_{i=1}^{n} \beta_{2i}(v))) \wedge (-\tilde{s}_{u}(\beta_{3})) \wedge \tilde{i}_{u}(\beta_{3})]$	$0 \lor (\lor_{i=1}^n \beta_{2i}(v))) \lor (-\tilde{i}_u(\beta_3)) \lor \tilde{s}_u(\beta_3)]$
L_{c}	$(-1) \lor [(-\tilde{s}_u(\beta_3)) \land (0 \land (\land_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \land (\land_{i=1}^n \beta_{2i}(v)) +$	$1 \wedge [(-\tilde{i}_u(\beta_3)) \vee (0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) + 0 \vee (\vee_{i=1}^n \beta_{2i}(v)) +$
10	$ ilde{i}_u(eta_3))]$	$ ilde{s}_u(eta_3))]$
I_7	$(-1) \lor [(0 \land (\land_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \land (\land_{i=1}^n \beta_{2i}(v))) \land (-0 \lor 2\tilde{s}_u(\beta_3) -$	$1 \wedge [(0 \lor (\lor_{i=1}^{n} \tilde{s}_{u}(\beta_{1i})) + 0 \lor (\lor_{i=1}^{n} \beta_{2i}(v))) \lor (-0 \land 2\tilde{i}_{u}(\beta_{3}) -$
	$ ilde{s}_u(eta_3^2))]$	$\widetilde{i}_u(eta_3^2))]$
I_8	$(-1) \vee [0 \wedge (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i})) + 0 \wedge (\wedge_{i=1}^{n} \beta_{2i}(v)) + 0 \wedge \tilde{i}_{u}(\beta_{3}) \wedge$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) + 0 \vee (\vee_{i=1}^n \beta_{2i}(v)) + 0 \vee \tilde{s}_u(\beta_3) \vee$
	$(- ilde{s}_u(eta_3))]$	$(- ilde{i}_u(eta_3))]$
I_9	$(-1) \vee [(0 \land (\land_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \land (\land_{i=1}^n \beta_{2i}(v))) \land \tilde{i}_u(\beta_3)]$	$1 \wedge \left[\left(0 \lor \left(\lor_{i=1}^{n} \tilde{s}_{u}(\beta_{1i}) \right) + 0 \lor \left(\lor_{i=1}^{n} \beta_{2i}(v) \right) \right) \lor \tilde{s}_{u}(\beta_{3}) \right]$
I_{10}	$(-1) \vee [0 \wedge (\wedge_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \wedge (\wedge_{i=1}^n \beta_{2i}(v)) + 0 \wedge \tilde{i}_u(\beta_3)]$	$1 \wedge [0 \lor (\lor_{i=1}^n \tilde{s}_u(\beta_{1i})) + 0 \lor (\lor_{i=1}^n \beta_{2i}(v)) + 0 \lor \tilde{s}_u(\beta_3)]$

Afterward, we look at the BKS-T algorithm. The corresponding solution comes in the form (6). We utilize the notation $\xi_n(u, v) = \{[C(u) + \beta_3(u)] \rightarrow \bigvee_{i=1}^n [(A_i(u) + \beta_{1i}(u)) \otimes (B_i(v) + \beta_{2i}(v))]\} - \{C(u) \rightarrow \bigvee_{i=1}^n [A_i(u) \otimes B_i(v)]\}.$

Theorems 12–14 are proved in the same manner.

Theorem 12: Assume that $A_i, C \in F(U)$ and $B_i \in F(V)$, and that β_{1i}, β_{2i} , and β_3 are the factors of perturbation of A_i, B_i , and C in turn (i = 1, 2, ..., n). If BKS-T is adopted, and \otimes takes \otimes_M , then the conclusions are provided in Table VI.

Theorem 13: Assume that $A_i, C \in F(U)$ and $B_i \in F(V)$, and that β_{1i}, β_{2i} , and β_3 are the factors of perturbation of A_i, B_i , and C in turn (i = 1, 2, ..., n). If BKS-T is used, and \otimes takes \otimes_P , then we have the outcomes in Table VII.

Theorem 14: Let $A_i, C \in F(U), B_i \in F(V)$, and $\beta_{1i}, \beta_{2i}, \beta_3$ be the factors of perturbation of A_i, B_i, C in turn

(i = 1, 2, ..., n). The BKS-T with \otimes takes \otimes_L yields the results in Table VIII.

Let $A_1, \ldots, A_n, C \in F(U)$, and $B_1, \ldots, B_n \in F(V)$. If there exists a small positive real number ε , and functions $\theta_{ki}(u) : U \to [-1, 1]$ for k = 1, 3 or $\theta_{2i}(v) : V \to [-1, 1]$ such that $\beta_{ki}(u) = \varepsilon \theta_{ki}(u)$ where k = 1, 3 and $\beta_{2i}(v) = \varepsilon \theta_{2i}(v)$ $(i = 1, 2, \ldots, n)$, then we obtain from Theorems 11–14 that $\lim_{\varepsilon \to 0} BKS(A_1^*, \ldots, A_n^*, B_1^*, \ldots, B_n^*, C^*)(v) = BKS(A_1, \ldots, A_n, B_1, \ldots, B_n, C)(v)$. Consequently, the BKS algorithm is stable on the strength of Definition 16-1).

From a different viewpoint, assume that there is a perturbation sequence $\{(A_{1m}^*, \ldots, A_{nm}^*, B_{1m}^*, \ldots, B_{nm}^*, C_m^*)\}$ with regard to $\{(\beta_{11m}(u), \ldots, \beta_{1nm}(u), \beta_{21m}(v), \ldots, \beta_{2nm}(v), \beta_{3m}(u))\}$, that is, $A_{1m}^*(u) = A_1(u) + \beta_{11m}(u), \ldots, A_{nm}^*(u) = A_n(u) + \beta_{1nm}(u), B_{1m}^*(v) = B_1(v) + \beta_{21m}(v), \ldots, B_{nm}^*(v) = B_n(v) + \beta_{2nm}(v), C_m^*(u) = C(u) + \beta_{3m}(u)$

TABLE VIII Estimation of Simple Perturbations for Multiple Rules Under BKS-T With \otimes_L (for Theorem 14)

\rightarrow	Lower bound of $\Delta_n(v)$	Upper bound of $\Delta_n(v)$
I_1	$(-1) \vee [0 \wedge (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i}) + \wedge_{i=1}^{n} \beta_{2i}(v)) \wedge (-\tilde{s}_{u}(\beta_{3}))]$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i}) + \vee_{i=1}^n \beta_{2i}(v)) \vee (-\tilde{i}_u(\beta_3))]$
I_2	$(-1) \vee [0 \wedge (\wedge_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \wedge (\wedge_{i=1}^n \beta_{2i}(v)) - 0 \vee \tilde{s}_u(\beta_3)]$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) + 0 \vee (\vee_{i=1}^n \beta_{2i}(v)) - 0 \wedge \tilde{i}_u(\beta_3)]$
I_3	$(-1) \vee [0 \wedge (\wedge_{i=1}^n \tilde{i}_u(\beta_{1i})) + 0 \wedge (\wedge_{i=1}^n \beta_{2i}(v)) + 0 \wedge (-\tilde{s}_u(\beta_3))]$	$1 \wedge [0 \vee (\vee_{i=1}^{n} \tilde{s}_{u}(\beta_{1i})) + 0 \vee (\vee_{i=1}^{n} \beta_{2i}(v)) + 0 \vee (-\tilde{i}_{u}(\beta_{3}))]$
I_4	$(-1) \vee [0 \wedge (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i}) + \wedge_{i=1}^{n} \beta_{2i}(v)) \wedge (-\tilde{s}_{u}(\beta_{3})) \wedge \tilde{i}_{u}(\beta_{3})]$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i}) + \vee_{i=1}^n \beta_{2i}(v)) \vee (-\tilde{i}_u(\beta_3)) \vee \tilde{s}_u(\beta_3)]$
Ι-	$(-1) \vee [0 \wedge (\wedge_{i=1}^n (-\tilde{s}_u(\beta_{1i}) - \beta_{2i}(v))) \wedge (\wedge_{i=1}^n \tilde{i}_u(\beta_{1i}) +$	$1 \wedge [0 \vee (\vee_{i=1}^n (-\tilde{i}_u(\beta_{1i}) - \beta_{2i}(v))) \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i}) +$
19	$\wedge_{i=1}^{n}\beta_{2i}(v))\wedge(-\tilde{s}_{u}(\beta_{3}))\wedge\tilde{i}_{u}(\beta_{3})]$	$\vee_{i=1}^{n}\beta_{2i}(v))\vee(-\tilde{i}_{u}(\beta_{3}))\vee\tilde{s}_{u}(\beta_{3})]$
I_6	$(-1) \lor [(-\tilde{s}_u(\beta_3)) \land (0 \land (\land_{i=1}^n \tilde{i}_u(\beta_{1i}) + \land_{i=1}^n \beta_{2i}(v)) + \tilde{i}_u(\beta_3))]$	$1 \wedge [(-\tilde{i}_u(\beta_3)) \vee (0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i}) + \vee_{i=1}^n \beta_{2i}(v)) + \tilde{s}_u(\beta_3))]$
<i>I</i>	$(-1) \vee [0 \wedge (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i}) + \wedge_{i=1}^{n} \beta_{2i}(v)) \wedge (-0 \vee 2\tilde{s}_{u}(\beta_{3}) -$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i}) + \vee_{i=1}^n \beta_{2i}(v)) \vee (-0 \wedge 2\tilde{i}_u(\beta_3) -$
17	$ ilde{s}_u(eta_3^2))]$	$ ilde{i}_u(eta_3^2))]$
I_{\circ}	$(-1) \vee [0 \wedge (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i})) + 0 \wedge (\wedge_{i=1}^{n} \beta_{2i}(v)) + 0 \wedge \tilde{i}_{u}(\beta_{3}) \wedge$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) + 0 \vee (\vee_{i=1}^n \beta_{2i}(v)) + 0 \vee \tilde{s}_u(\beta_3) \vee$
18	$(- ilde{s}_u(eta_3))]$	$(- ilde{i}_u(eta_3))]$
I_9	$(-1) \vee [0 \wedge (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i}) + \wedge_{i=1}^{n} \beta_{2i}(v)) \wedge \tilde{i}_{u}(\beta_{3})]$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i}) + \vee_{i=1}^n \beta_{2i}(v)) \vee \tilde{s}_u(\beta_3)]$
I_{10}	$(-1) \vee [0 \wedge (\wedge_{i=1}^{n} \tilde{i}_{u}(\beta_{1i})) + 0 \wedge (\wedge_{i=1}^{n} \beta_{2i}(v)) + 0 \wedge \tilde{i}_{u}(\beta_{3}) \wedge$	$1 \wedge [0 \vee (\vee_{i=1}^n \tilde{s}_u(\beta_{1i})) + 0 \vee (\vee_{i=1}^n \beta_{2i}(v)) + 0 \vee \tilde{s}_u(\beta_3) \vee$
110	$(- ilde{s}_u(eta_3))]$	$(- ilde{i}_u(eta_3))]$

 $(m = 1, 2, \ldots). \text{ If } \lim_{m \to \infty} \sup_{u \in U} |\beta_{11m}(u)| = \cdots = \lim_{m \to \infty} \sup_{u \in U} |\beta_{1nm}(u)| = 0, \quad \lim_{m \to \infty} \sup_{v \in V} |\beta_{21m}(v)| = \dots = \lim_{m \to \infty} \sup_{v \in V} |\beta_{2nm}(v)| = 0, \quad \lim_{m \to \infty} \sup_{u \in U} |\beta_{3m}(u)| = 0 \text{ hold, then we gain from Theorems 11-14} that \quad \lim_{m \to \infty} BKS(A_{1m}^*, \ldots, A_{nm}^*, B_{1m}^*, \ldots, B_{nm}^*, C_m^*)(v) = BKS(A_1, \ldots, A_n, B_1, \ldots, B_n, C)(v).$

In sum, the BKS algorithm for the situation of multiple rules is stable from the perspective of simple perturbation.

Affective computing [37] is nowadays one of the most active research areas. Emotion deduction (which explores how to generate reasonable values of other emotions from some basic emotions) makes a critical difference in many aspects (e.g., when constructing massive emotional corpus, affect state transitions, and so forth), which is a vital task of affective computing. In what follows, we offer two examples of emotion deduction.

Example 1: We carry computing of emotion aspects via the BKS-I method where \rightarrow is implemented as I_4 . For the eight basic emotions, (including surprise, expect, anxiety, sorrow, angry, hate, joy, and love), we find that the former six emotions have a strong relationship with fear (as a new emotion). We establish the emotion deduction system from six basic emotions to fear. In detail, let $U = \{u_1, u_2, \dots, u_6\}$ where $u_1 = 0, u_2 = 0.2, u_3 = 0.4, u_4 = 0.6, u_5 = 0.8, u_6 = 1.0, and <math>V = \{v_1\}$ where $v_1 = 0.5$. Meanwhile, some rules from A_i to B_i and an input *C* are as follows:

$$A_{1} = \frac{0.3}{u_{1}} + \frac{0.9}{u_{2}} + \frac{0.2}{u_{3}} + \frac{0.5}{u_{4}} + \frac{0.3}{u_{5}} + \frac{0.4}{u_{6}}, \quad B_{1} = \frac{0.2}{v_{1}}$$

$$A_{2} = \frac{0.5}{u_{1}} + \frac{0.6}{u_{2}} + \frac{0.5}{u_{3}} + \frac{0.7}{u_{4}} + \frac{0.6}{u_{5}} + \frac{0.7}{u_{6}}, \quad B_{2} = \frac{0.4}{v_{1}}$$

$$A_{3} = \frac{0.9}{u_{1}} + \frac{0.4}{u_{2}} + \frac{0.7}{u_{3}} + \frac{0.9}{u_{4}} + \frac{0.5}{u_{5}} + \frac{0.8}{u_{6}}, \quad B_{3} = \frac{0.6}{v_{1}}$$

$$A_{4} = \frac{0.6}{u_{1}} + \frac{0.7}{u_{2}} + \frac{0.9}{u_{3}} + \frac{0.4}{u_{4}} + \frac{0.3}{u_{5}} + \frac{0.5}{u_{6}}, \quad B_{4} = \frac{0.9}{v_{1}}$$

$$C = \frac{0.8}{u_{1}} + \frac{0.7}{u_{2}} + \frac{0.6}{u_{3}} + \frac{0.9}{u_{4}} + \frac{0.5}{u_{5}} + \frac{0.8}{u_{6}}.$$

Here, the input A_i reflects the value for six basic emotions and the output B_i stands for the value for the emotion fear. For the input *C*, the BKS-I solution from (7) is $D(v_1) =$ $\inf_{u \in U} \{ C(u) \to \bigwedge_{i=1}^n (A_i(u) \to B_i(v_1)) \} = 0.5 \land 0.3 \land 0.5 \land 0.4 \land 0.5 \land 0.4 = 0.3.$

For one situation, we take $\beta_{11}(u) = 0.05$, $\beta_{12}(u) = 0.06$, $\beta_{13}(u) = 0.07$, $\beta_{14}(u) = 0.08$, $\beta_{21}(v) = 0.06$, $\beta_{22}(v) = 0.07$, $\beta_{23}(v) = 0.08$, $\beta_{24}(v) = 0.09$, $\beta_{3}(u) = 0.05$ ($u \in U, v \in V$). Let $A_{i}^{*}(u) = A_{i}(u) + \beta_{1i}(u)$, $B_{i}^{*}(v) = B_{i}(v) + \beta_{2i}(v)$, and $C^{*}(u) = C(u) + \beta_{3}(u)$ (i = 1, 2, 3, 4).

From Theorem 11 for I_4 , we obtain that the lower bound of $\Delta_n(v_1)$ is $[\wedge_{i=1}^n (-\tilde{s}_u(\beta_{1i}))] \wedge [\wedge_{i=1}^n \tilde{i}_u(\beta_{1i})] \wedge [\wedge_{i=1}^n \beta_{2i}(v_1)] \wedge (-\tilde{s}_u(\beta_3)) \wedge \tilde{i}_u(\beta_3) = (-0.08) \wedge 0.05 \wedge 0.06 \wedge (-0.05) \wedge 0.05 = -0.08$. Analogously, the upper bound of $\Delta_n(v_1)$ is 0.09. Hence, we obtain $-0.08 \leq \Delta_n(v_1) \leq 0.09$ and $0.22 \leq BKS(A_1^*, \ldots, A_4^*, B_1^*, \ldots, B_4^*, C^*)$ $(v) \leq 0.39$.

For another situation, we adopt $\beta_{1i}(u) = \beta_{2i}(v) = \beta_3(u) =$ 0.02 $(u \in U, v \in V, i = 1, 2, 3, 4)$. Afterward, we can similarly gain $-0.02 \leq \Delta_n(v_1) \leq 0.02$ and $0.28 \leq$ $BKS(A_1^*, \ldots, A_4^*, B_1^*, \ldots, B_4^*, C^*)(v) \leq 0.32$. Obviously, in this case, the value of $BKS(A_1^*, \ldots, A_4^*, B_1^*, \ldots, B_4^*, C^*)(v)$ is closer to $BKS(A_1, \ldots, A_4, B_1, \ldots, B_4, C)(v)$.

Example 2: We involve the BKS-T method in which \rightarrow is realized as I_7 and \otimes is taken as \otimes_M . We use the same $A_1, \ldots, A_4, B_1, \ldots, B_4$ as Example 1. New input is $C = (0.7/u_1) + (0.2/u_2) + (0.3/u_3) + (0.8/u_4) + (0.9/u_5) + (0.6/u_6)$. Then, the BKS-T solution from (6) is $D(v_1) = \inf_{u \in U} \{C(u) \rightarrow \bigvee_{i=1}^n (A_i(u) \otimes B_i(v_1))\} = 0.6 \land 0.96 \land 0.91 \land 0.6 \land 0.5 \land 0.64 = 0.5$.

For one case, we choose $\beta_{11}(u) = 0.08$, $\beta_{12}(u) = 0.07$, $\beta_{13}(u) = 0.06$, $\beta_{14}(u) = 0.07$, $\beta_{21}(v) = 0.07$, $\beta_{22}(v) = 0.06$, $\beta_{23}(v) = 0.06$, $\beta_{24}(v) = 0.05$, and $\beta_{3}(u) = 0.06$ ($u \in U, v \in V$). Let $A_{i}^{*}(u) = A_{i}(u) + \beta_{1i}(u)$, $B_{i}^{*}(v) = B_{i}(v) + \beta_{2i}(v)$, and $C^{*}(u) = C(u) + \beta_{3}(u)$ (i = 1, 2, 3, 4).

In the light of Theorem 12 for I_7 , we have that the lower bound of $\Delta_n(v_1)$ is $(-1) \vee [(\wedge_{i=1}^n \tilde{i}_u(\beta_{1i})) \wedge (\wedge_{i=1}^n \beta_{2i}(v_1)) \wedge$ $(-0 \lor 2\tilde{s}_u(\beta_3) - \tilde{s}_u(\beta_3^2))] = (-1) \lor [0.06 \land 0.05 \land (-0 \lor 0.12 - 0.06^2)] = -0.1236$. Analogously the upper bound of $\Delta_n(v_1)$ is 0.08. Consequently, we find $-0.1236 \le \Delta_n(v_1) \le 0.08$ and $0.3764 \le BKS(A_1^*, \dots, A_4^*, B_1^*, \dots, B_4^*, C^*)(v) \le 0.58$.

For another case, we let $\beta_{1i}(u) = \beta_{2i}(v) = \beta_3(u) = 0.02$ ($u \in U, v \in V, i = 1, 2, 3, 4$). In a similar way we obtain $-0.0404 \leq \Delta_n(v_1) \leq 0.02$ and $0.4596 \leq BKS(A_1^*, \dots, A_4^*, B_1^*, \dots, B_4^*, C^*)(v) \leq 0.52$. In this case, $BKS(A_1^*, \dots, A_4^*, B_1^*, \dots, B_4^*, C^*)(v)$ evidently achieves a closer result to $BKS(A_1, \dots, A_4, B_1, \dots, B_4, C)(v)$.

VI. DISCUSSION

- Definition 9 (i.e., Pappis's proximity measure [17]), Definition 10 (i.e., Hong and Hwang' ε-similarity [18]), and Definition 12 (viz., Cai's δ-equality [34]) can be incorporated into the framework of interval perturbation of fuzzy sets. Moreover, Definition 11 (i.e., Ying's maximum ε perturbation of a fuzzy set in [19]) can be deemed as a special situation of simple perturbation. Then, the interval perturbation and simple perturbation are more general ways to express the robust issue. So this work is more meaningful vis-à-vis the previous studies.
- 2) In this study, we obtain the upper and lower bounds of BKS output deviation derived from the perturbation of the input fuzzy set, which provide more detailed characterization for the output deviation along with the input perturbation.
- 3) The previous works have exhibited that the BKS method has good performance, including the interpolativity, continuity, robustness, and that the BKS-based FRI is as effective and efficient as the CRI method (see [7], [12], and [13]). As their development, this work goes a step further to validate the stability of BKS from the perspective of oscillation-bound estimation of perturbations, which is also as good as the CRI method.
- 4) In [29], the CRI-based FRI with varying limits of input is researched. Our work in this study for the BKS-based FRI has three disparate and original points with [29] as follows.
 - a) The work in [29] only focuses on modeling form \check{R} in CRI, where the CRI solution is $D(v) = \sup_{u \in U} \{C(u) \otimes R(u, v)\} = \sup_{u \in U} \{C(u) \otimes (A(u) \rightarrow B(v))\} (v \in V)$. Thereinto, $\check{R}_1(u, v) = A(u) \rightarrow B(v)$ is adopted. However, in this study, we discover the situations of both \check{R} (using \rightarrow) and \hat{R} (employing \otimes) in BKS. Hence, our modeling strategies are more comprehensive.
 - b) The work in [29] only aims at one rule in CRI. That is, $\check{R}_1(u, v) = A(u) \rightarrow B(v)$ is considered. By our findings in this study, the performance of BKS is as good as CRI for the situation of \check{R}_1 . But more importantly, we make a thorough inquiry into the cases of both one rule and multiple rules for BKS. To be specific, four kinds of situations are revealed, which incorporate $\check{R}_1(u, v) =$ $A(u) \rightarrow B(v), \, \hat{R}_2(u, v) = A(u) \otimes B(v), \, \check{R}_3(u, v) =$

 $\vee_{i=1}^{n} \{A_i(u) \otimes B_i(v)\}, \text{ and } \hat{R}_4(u, v) = \wedge_{i=1}^{n} \{A_i(u) \rightarrow B_i(v)\} \ (u \in U, v \in V). \text{ Notice that the results of BKS for the last three are also good.}$

c) The chain of fuzzy reasoning has the problem of error propagation since it consists of multiple nested inferences. Consequently, it has a stricter demand for stability and possesses more important value for the stability research than the ordinary fuzzy reasoning. The work in [29] does not consider the problem of the fuzzy reasoning chain. However, this study explores such challenging problem and has a confirmed the stability of BKS chain reasoning.

VII. CONCLUSION

It should be stressed that the previous works mainly concentrated on how the output values were altered due to perturbation parameters of input values. However, studies on estimating oscillation bounds of output values with respect to varying limits of input values were not available. This study directed at this problem and offered the upper and lower bounds of BKS output, in which the two bounds are characterized in the form of fuzzy sets. Then, the stable (or asymptotically stable) properties are verified for the BKS algorithms. We systematically investigated the oscillation-bound estimation of perturbations for the fuzzy reasoning method of BKS. The key contributions are as follows.

Above all, we focus on two kinds of BKS algorithms, (including the BKS-T and BKS-I algorithms corresponding to two different modeling methods for the rule base) with one rule as well as multiple rules, and research the situation of interval perturbation. Corresponding upper and lower bounds of BKS output variation are offered, where the *R*-implication, (*S*, *N*)-implication, QL-implication, and *t*-norm implication are adopted. The stability of the BKS method for interval perturbation has been validated.

Furthermore, focusing on the chain of fuzzy reasoning with BKS, the BKS output scope originating from input interval perturbation are given, where the *R*-implication, (*S*, N)-implication, QL-implication, and *t*-norm implication are utilized. The stability of BKS chain reasoning has been confirmed.

In the end, we investigated the situation of simple perturbation for the BKS algorithm with one rule and multiple rules and obtained the upper and lower bounds of BKS output deviation derived from the simple perturbation of the input fuzzy set, in which ten fuzzy implications, (including some representative *R*-implications) (*S*, *N*)-implications, QL-implications, and *t*-norm implications) together with three *t*-norms are employed. The stable and asymptotic stable properties of these BKS strategies are verified.

It is noted that the interval perturbation and simple perturbation are more general expression ways for the robust issue, hence the work for the interval perturbation and simple perturbation is more meaningful than the previous works. In addition, the obtained oscillation bound gives more detailed characterization for the output deviation along with the input perturbation. This work goes a step further to validate the good properties of the BKS method of fuzzy reasoning, which is as good as the CRI method.

Here, we emphasize the novelty of this study.

- The oscillation-bound estimation of perturbations is a novel problem to be explored for BKS (noting that the foregoing works in relation to stability and robustness of BKS mainly focused on how the output values were altered due to perturbation parameters of input values).
- Directing at fuzzy chain reasoning with BKS, its output scope resulting from input interval perturbation is offered. Note that such a point for the chain of FRI was lacking in the previous research.
- 3) We explore the error estimation of perturbations for both \check{R} and \hat{R} in the BKS scheme. It is noticed that such a topic for the situation of \hat{R} was not investigated in the foregoing studies on FRI.

In future studies, it is worth developing the BKS method from the perspective of granular computing (see [38] and [39]) and studying on how to construct and design rational fuzzy controllers by choosing some ideal BKS strategies.

REFERENCES

- [1] K. K. Ang, C. Quek, and M. Pasquier, "POPFNN-CRI(S): Pseudo outer product based fuzzy neural network using the compositional rule of inference and singleton fuzzifier," *IEEE Trans. Syst., Man, Cybern. B, Cybern.*, vol. 33, no. 6, pp. 838–849, Dec. 2003.
- [2] D. S. Yeung and E. C. C. Tsang, "A comparative study on similaritybased fuzzy reasoning methods," *IEEE Trans. Syst., Man, Cybern. B, Cybern.*, vol. 27, no. 2, pp. 216–227, Apr. 1997.
- [3] J. F. Baldwin, "A new approach to approximate reasoning using a fuzzy logic," *Fuzzy Sets Syst.*, vol. 2, no. 4, pp. 309–325, Oct. 1979.
- [4] S.-M. Chen and W.-C. Hsin, "Weighted fuzzy interpolative reasoning based on the slopes of fuzzy sets and particle swarm optimization techniques," *IEEE Trans. Cybern.*, vol. 45, no. 7, pp. 1250–1261, Jul. 2015.
- [5] Y. M. Tang and W. Pedrycz, "On the α(u,v)-symmetric implicational method for R-and (S, N)-implications," *Int. J. Approx. Reason.*, vol. 92, pp. 212–231, Jan. 2018.
- [6] L. A. Zadeh, "Outline of a new approach to the analysis of complex systems and decision processes," *IEEE Trans. Syst., Man, Cybern., Syst.*, vol. SMC-3, no. 1, pp. 28–44, Jan. 1973.
- [7] W. Pedrycz, "Applications of fuzzy relational equations for methods of reasoning in presence of fuzzy data," *Fuzzy Sets Syst.*, vol. 16, no. 2, pp. 163–175, 1985.
- [8] W. Bandler and L. J. Kohout, "Fuzzy relational products and fuzzy implication operators," presented at the Int. Workshop Fuzzy Reason. Theory Appl., London, U.K., 1978, pp. 239–244.
- [9] D. Dubois, H. Prade, and L. Ughetto, "Checking the coherence and redundancy of fuzzy knowledge bases," *IEEE Trans. Fuzzy Syst.*, vol. 5, no. 3, pp. 398–417, Aug. 1997.
- [10] E. P. Klement, R. Mesiar, and E. Pap, *Triangular Norms*. Dordrecht, The Netherlands: Kluwer Acad. Pub., 2000.
- [11] D. Dubois and H. Prade, "What are fuzzy rules and how to use them," *Fuzzy Sets Syst.*, vol. 84, no. 2, pp. 169–185, Dec. 1996.
- [12] M. Štěpnička and B. Jayaram "On the suitability of the Bandler-Kohout subproduct as an inference mechanism," *IEEE Trans. Fuzzy Syst.*, vol. 18, no. 2, pp. 285–298, Apr. 2010.

- [13] S. Mandal and B. Jayaram, "Bandler-Kohout subproduct with Yager's classes of fuzzy implications," *IEEE Trans. Fuzzy Syst.*, vol. 22, no. 3, pp. 469–482, Jun. 2014.
- [14] M. Štěpnička and S. Mandal, "Fuzzy inference systems preserving Moser-Navara axioms," *Fuzzy Set Syst.*, vol. 338, pp. 97–116, May 2018.
- [15] U. Bodenhofer, M. Dankova, M. Štěpnička, and V. Novak, "A plea for the usefulness of the deductive interpretation of fuzzy rules in engineering applications," in *Proc. IEEE Int. Conf. Fuzzy Syst.*, London, U.K., 2007, pp. 1567–1572.
- [16] L.-X. Wang, A Course in Fuzzy Systems and Control. Englewood Cliffs, NJ, USA: Prentice-Hall, 1997.
- [17] C. P. Pappis, "Value approximation of fuzzy systems variables," *Fuzzy Sets Syst.*, vol. 39, no. 1, pp. 111–115, Jan. 1991.
- [18] D. H. Hong and S. Y. Hwang, "A note on the value similarity of fuzzy systems variables," *Fuzzy Sets Syst.*, vol. 66, no. 3, pp. 383–386, Sep. 1994.
- [19] M. S. Ying, "Perturbation of fuzzy reasoning," *IEEE Trans. Fuzzy Syst.*, vol. 7, no. 5, pp. 625–629, Oct. 1999.
- [20] K.-Y. Cai, "Robustness of fuzzy reasoning and δ-equalities of fuzzy sets," *IEEE Trans. Fuzzy Syst.*, vol. 9, no. 5, pp. 738–750, Oct. 2001.
- [21] J. H. Jin, Y. M. Li, and C. Q. Li, "Robustness of fuzzy reasoning via logically equivalence measure," *Inf. Sci.*, vol. 177, no. 22, pp. 5103–5117, Nov. 2007.
- [22] S. S. Dai, D. W. Pei, and S.-M. Wang, "Perturbation of fuzzy sets and fuzzy reasoning based on normalized Minkowski distances," *Fuzzy Sets Syst.*, vol. 189, no. 1, pp. 63–73, Feb. 2012.
- [23] Y. F. Li, K. Y. Qin, and X. X. He, "Robustness of fuzzy connectives and fuzzy reasoning," *Fuzzy Sets Syst.*, vol. 225, pp. 93–105, Aug. 2013.
- [24] Y. F. Li, K. Y. Qin, X. X. He, and D. Meng, "Robustness of fuzzy connectives and fuzzy reasoning with respect to general divergence measures," *Fuzzy Sets Syst.*, vol. 294, pp. 63–78, Jul. 2016.
- [25] Y. F. Li, K. Y. Qin, and X. X. He, "Dissimilarity functions and divergence measures between fuzzy sets," *Inf. Sci.*, vol. 288, pp. 15–26, Dec. 2014.
- [26] F. Klawonn and J. L. Castro, "Similarity in fuzzy reasoning," *Mathware Soft Comput.*, vol. 2, no. 3, pp. 197–228, 1995.
- [27] M. Štěpnička, B. Jayaram, and Y. Su, "A short note on fuzzy relational inference systems," *Fuzzy Set Syst.*, vol. 338, pp. 90–96, May 2018.
- [28] B. Moser and M. Navara, "Fuzzy controllers with conditionally firing rules," *IEEE Trans. Fuzzy Syst.*, vol. 10, no. 3, pp. 340–348, Jun. 2002.
- [29] G. S. Cheng and Y. X. Fu, "Error estimation of perturbations under CRI," *IEEE Trans. Fuzzy Syst.*, vol. 14, no. 6, pp. 709–715, Dec. 2006.
- [30] M. Mas, M. Monserrat, J. Torrens, and E. Trillas, "A survey on fuzzy implication functions," *IEEE Trans. Fuzzy Syst.*, vol. 15, no. 6, pp. 1107–1121, Dec. 2007.
- [31] J. Fodor and M. Roubens, Fuzzy Preference Modeling and Multicriteria Decision Support. Dordrecht, The Netherlands: Kluwer Acad. Pub., 1994.
- [32] M. Baczyński and B. Jayaram, "QL-implications: Some properties and intersections," *Fuzzy Set Syst.*, vol. 161, no. 2, pp. 158–188, Jan. 2010.
- [33] C. Carlsson, R. Fullér, and S. Fullér, "Possibility and necessity in weighted aggregation," in *The Ordered Weighted Averaging Operators: Theory, Methodology, and Applications.* R.R. Yager and J. Kacprzyk, Eds. Boston, MA, USA: Kluwer, 1997, pp. 18–28.
- Eds. Boston, MA, USA: Kluwer, 1997, pp. 18–28.
 [34] K.-Y. Cai, "δ-equalities of fuzzy sets," *Fuzzy Set Syst.*, vol. 76, no. 1, pp. 97–112, Nov. 1995.
- [35] M. S. Ying, "Implication operators in fuzzy logic," *IEEE Trans. Fuzzy Syst.*, vol. 10, no. 1, pp. 88–91, Feb. 2002.
- [36] M. Baczyński and B. Jayaram, "(S, N)- and R-implications: A stateof-the-art survey," *Fuzzy Set Syst.*, vol. 159, no. 14, pp. 1836–1859, Jul. 2008.
- [37] H. Y. Meng and N. Bianchi-Berthouze, "Affective state level recognition in naturalistic facial and vocal expressions," *IEEE Trans. Cybern.*, vol. 44, no. 3, pp. 315–328, Mar. 2014.
- [38] W. Pedrycz, Granular Computing: Analysis and Design of Intelligent Systems. Boca Raton, FL, USA: CRC Press/Francis & Taylor, 2013.
- [39] Y. H. Shen, W. Pedrycz, and X. M. Wang, "Clustering homogeneous granular data: Formation and evaluation," *IEEE Trans. Cybern.*, vol. 49, no. 4, pp. 1391–1402, Apr. 2019.



Yiming Tang (Member, IEEE) received the Ph.D. degree from the Hefei University of Technology, Hefei, China, in 2011.

He is an Associate Professor with the Hefei University of Technology. He is a Visiting Professor with the University of Alberta, Edmonton, AB, Canada. He has published more than 60 papers. His research interests include fuzzy control, clustering, image processing, and affective computing.

Dr. Tang is an Associate Editor of IEEE ACCESS, a Member of Editorial Board of *PLOS ONE*, a

Member of Editorial Board of *Mathematics and Computer Science*, and an Associate Editor of *Journal of Mathematics and Informatics*, and also a Senior Member of the Chinese Association for Artificial Intelligence (CAAI) and a Senior Member of China Computer Federation. He serves as a Professional Committee of Granular Computing and Knowledge Discovery of CAAI, a Professional Committee of Machine Learning of CAAI. He is a Reviewer of more than 20 journals.



Witold Pedrycz (Life Fellow, IEEE) received the M.Sc., Ph.D., and D.Sc. degrees from the Silesian University of Technology, Gliwice, Poland, in 1977, 1980, and 1984, respectively.

He is a Professor and the Canada Research Chair of Computational Intelligence with the Department of Electrical and Computer Engineering, University of Alberta, Edmonton, AB, Canada. He is also with the Systems Research Institute, Polish Academy of Sciences, Warsaw, Poland, where he was elected as a Foreign Member in 2009. He also holds an appoint-

ment as a Special Professor with the School of Computer Science, University of Nottingham, Nottingham, U.K. He has authored 14 research monographs that cover various aspects of computational intelligence and software engineering. He has edited a number of volumes. His current research interests include computational intelligence, fuzzy modeling and granular computing, knowledge discovery and data mining, fuzzy control, and software engineering.

Prof. Pedrycz received the prestigious Norbert Wiener Award from the IEEE Systems, Man, and Cybernetics Council in 2007, the IEEE Canada Computer Engineering Medal in 2008, and the Cajastur Prize for Soft Computing in 2009 from the European Centre for Soft Computing for pioneering and multifaceted contributions to granular computing. He is an Editor-in-Chief of *Information Sciences* and *International Journal of Granular Computing*. He is also an Associate Editor of IEEE TRANSACTIONS ON FUZZY SYSTEMS and IEEE TRANSACTIONS ON SYSTEMS, MAN, AND CYBERNETICS: SYSTEMS. He has been a member of numerous program committees of IEEE conferences in the area of fuzzy sets and neurocomputing. He is a Fellow of the Royal Society of Canada.